

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Mocanu, Tatiana

Working Paper Designing Gender Equity: Evidence from Hiring Practices

IZA Discussion Papers, No. 17480

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Mocanu, Tatiana (2024) : Designing Gender Equity: Evidence from Hiring Practices, IZA Discussion Papers, No. 17480, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/308339

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 17480

Designing Gender Equity: Evidence from Hiring Practices

Tatiana Mocanu

NOVEMBER 2024



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 17480

Designing Gender Equity: Evidence from Hiring Practices

Tatiana Mocanu Columbia University and IZA

NOVEMBER 2024

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

Designing Gender Equity: Evidence from Hiring Practices^{*}

I combine novel data on job applications and hiring decisions for the universe of public sector jobs in Brazil and a natural experiment that decreased discretion in hiring to analyze how screening determines gender application and hiring gaps. I find that hiring practices have crucial gender equity consequences for selection and sorting, and not all approaches to reduce discretion have the same implications. Limiting discretion in existing tools or adding new impartial tools reduces the gender hiring gap by a third. However, policies that eliminate subjective tools like interviews are ineffective, suggesting employers should carefully weigh bias-information trade-offs.

JEL Classification:	M51, D73, J16, J45, J71
Keywords:	hiring practices, gender bias, public sector personnel

Corresponding author:

Tatiana Mocanu Columbia University 116th and Broadway New York, NY 10027 USA E-mail: tm3326@columbia.edu

^{*} I am deeply grateful for the guidance and support from David Albouy, Alex Bartik, and Dan Bernhardt. This paper has benefited greatly from feedback and discussions with Vittorio Bassi, Barbara Biasi, Nicola Bianchi, Sandra Black, Eliza Forsythe, Yana Gallen, Andy Garin, Pat Kline, Dmitri Koustas, Gianmarco León-Ciliotta, Day Manoli, Sendhil Mullainathan, Kieu-Trang Nguyen, Matthew Notowidigdo, Dan-Olof Rooth, Matteo Paradisi, Nicola Persico, Brendan Price, Simon Quinn, Natalia Rigol, Olga Stoddard, Pedro Tremacoldi-Rossi, Sergio Urzua, Chris Walters, Russell Weinstein, Guo Xu, Owen Zidar, as well as seminar and conference participants at Columbia, UC Berkeley, Northwestern Kellogg, Cornell, Boston University, Oxford, Bocconi, USC Marshall, Princeton, EIEF, CEMFI, University of Virginia Batten, Stockholm University, UCL, Rutgers, University of Illinois Urbana-Champaign, Sciences Po Summer Workshop: Labor, NBER Summer Institute Gender in the Economy, Discrimination in the 21st Century Conference (University of Chicago), Stanford Institute for Theoretical Economics Gender, Emerging Scholars Conference (University of Wisconsin-Madison), CESifo Area Conference on Labor Economics, Call to Service Conference (Stanford University), SOLE, Annual Congress of the International Institute of Public Finance (Utah State University), North America Summer Meeting of the Econometric Society, CESifo/ifo Junior Workshop on Big Data, ERMAS Applied Monthly Seminar, and the 3rd Monash-Warwick-Zurich Text-as-Data Workshop. I also thank several career public servants in various Brazilian government levels for sharing their experience in participating and conducting hiring processes in the country's public sector.

1 Introduction

Much progress has been made in the last decades toward closing gender gaps in labor market outcomes. Despite that, women continue to face barriers to make inroads in many careers, putting into focus the equity consequences of how firms and organizations recruit and select job applicants. While the existence of discrimination in hiring has been well documented,¹ little is known about the extent to which different screening practices may mitigate or exacerbate gender disparities.

How to best design hiring processes that are less susceptible to bias is a question that remains open. Employers have at their disposal both subjective and objective screening tools when making hiring decisions. On one hand, subjective tools, such as interviews, that give evaluators more discretion may provide additional productivity information, but are more prone to evaluator biases. On the other hand, more objective tools, such as written tests, constrain evaluator behavior but may screen out otherwise qualified workers.

These different properties of screening tools imply that alternative ways to reduce discretion may have distinct consequences for diversity. First, the choice of screening practices may impose a bias-information trade-off. Removal of subjective steps discards information about a candidate that is difficult to learn otherwise. And removing information about minority applicants, even when biased, may not necessarily be effective since evaluators may instead rely more heavily on priors about group characteristics. Second, introducing objective tools, while constraining evaluator behavior, could inadvertently introduce disparate impact.² And to make things even more complicated, changes in screening tools can also shift whether women apply for a job in the first place.

Using Brazil's public sector as setting, this paper studies the role of screening methods in determining gender application and hiring gaps. Tackling these questions empirically has been challenging because hiring decisions are effectively a black box. Economists rarely observe how organizations screen candidates and make selection decisions. In addition, since employers' choices of how to screen their employees is endogenous, exogenous variation necessary for causal analysis has limited empirical work. As Oyer and Schaefer (2011) put it: "What manager, after all, would allow an academic economist to experiment with the firm's screening, interviewing or hiring decisions?".

¹See, for example, Goldin and Rouse (2000), Bertrand and Mullainathan (2004), Kline et al. (2022).

²The standard definition for disparate impact since Griggs v. Duke Power Co. (1971) is that, even if discrimination is unintentional, a job requirement that leads to disparities between applicants from different groups can only be legally justified if it connects to job duties. A prominent example in the US is the use of aptitude test in police hiring, on which black applicants perform worse white applicants, but the entrance examination scores are not predictive of performance on the job (McCrary (2007)).

I combine novel data on job applications and hiring decisions for the universe of public sector jobs and a natural experiment that allow me to overcome these challenges. To open the black box of hiring, I develop a two-step, flexible natural language processing algorithm that distills over 35 million official government text files into data. These documents contain detailed records of the universe of Brazilian government screening processes, but have remained unavailable to researchers due to their highly confounding and unstructured nature. My novel query-based based algorithm generates a rich database detailing job applicant performance and evaluator decision making process, including job openings and offers, applicant and evaluator identities, and candidates' individual scores by screening method.

The paper leverages variation generated by the introduction of a requirement to public sector hiring mandating impartial screening. Public employees in Brazil are hired through a formal, merit-based selection system that has been in place for decades. With the passage of the country's new Constitution in 1988, parts of the constitutional text were modified to align with a new social focus. Concerning the selection of public servants, this came in the form of codifying the understanding that hiring practices that could enable differential treatment would be legally challenged, which led to a redesign of hiring processes toward a reduction in discretion.

The impartiality requirement was introduced without specific compliance directions. This feature generated a number of different changes to screening practices, with a treatment assignment process determined by the occupation. This followed the tradition of organizing rules related to public servants around careers, which in turn drove the discussions about how to modify hiring practices. Bureaucrats and legal experts at high organizational levels took into consideration the pre-existing screening tools in an occupation and modified selection processes in one of two general ways: make tools already in use more impartial or change the screening tools. While the first case amounted to fully or partially blinding written exams, the second included cases when written exams were added and non-written exams were removed or replaced. This richness in variation provides a unique empirical setting to study the equity consequences of alternative ways to reduce discretion.

In line with the legal interpretation provided by bureaucrats, 95% of hiring processes within a job title adopted only one change in screening methods. Yet, each treatment involved a variety of occupations across several dimensions. For example, kitchen assistants and civil engineers had screening modified in the same way, just like teachers and police officers were assigned the same treatment. Importantly, the introduction of the impartiality requirement did not attend specific policy goals, like increasing diversity, nor comprised a broader anti-discrimination effort, which I support empirically by leveraging thousands of carefully documented records about the drafting of Brazil's new Constitution.

My empirical analysis first documents the aggregate effects on gender gaps using a tripledifference design. I use an institutional delay of several years in adoption by states, in contrast with the sharp implementation by federal jobs, to compare outcomes of women and men in job processes within the same occupation in federal and states postings. The parallel trends assumption rules out time-varying unobserved variables that correlate with gender gaps and that could change differentially between state and federal jobs. Consistent with that, there were no contemporaneous changes to labor markets affecting treatment relative to control, and no pre-trends in gender gaps.³

To leverage the fact that my data allows to observe job seekers' applications and employers' hiring decisions, I decompose the components that determine the share of female hires in a hiring process — selection (demand) and sorting (supply) and estimate each effect separately. While the share of female hires captures whether the workforce is becoming more diverse and is usually the outcome observable to researchers, measuring demand and supply effects individually is important to understand how each margin adjusts to changes in the hiring process design, since they have distinct potential remedies and policy consequences.⁴

I estimate that women's final scores increased by 0.07 standard deviation, accompanied by a decrease of similar magnitude in men's scores. This resulted in the overall gender score gap closing in response to the reduction of hiring discretion. Women's final scores increase was driven by higher marks in written exams, which universally became blind. When still used, non-written stages had gender score gaps unchanged, suggesting evaluators did not respond strategically by favoring men in more discretionary practices. The decrease in the gender final score gap translates into an improved conditional probability of being hired for women and a narrowing of the gender hiring gap of about 44% of the pre-treatment level, even after controlling for job process competitiveness.

Who is hired ultimately depends not only on the employer's hiring decision, but also on who applies to begin with. Men and women may respond differently to perceived discrimination associated with unfair hiring practices or more costly preparation as selection processes are modified. Using the entire pool of applicants, including those rejected, I find that application rates of women relative to men increased by about 1 percentage point. Taken together, both higher employer demand and application rates of women result in the increase of gender diversity among hires just a few years after the new Constitution came into effect.

³I further support the credibility of the design underpinning my aggregate estimates through several checks. These include whether establishments tried to front-run their legislative process that required a long delay to implement the impartiality requirement for states (anticipation), or if the treatment assignment to federal jobs led to spillovers to control job processes. The weight of the evidence rejects the plausibility of these violations.

⁴This distinction is for example crucial to measure discrimination, since outcomes constructed from a sample of hired workers (e.g. share of female hires) are conditional on application decisions, which are endogenous (Durlauf and Heckman (2020)).

My initial set of results reveal that even within hiring systems with structured and professionalized selection procedures, discretion in hiring can still prevent women's access to jobs. This is particularly important in the context public bureaucracies, where having a more representative body of civil servants is linked to broader benefits to state capacity (Kingsley (1944), Alsan et al. (2019), Miller and Segal (2019), Xu (2021)), and the establishment of meritocratic recruitment alone, as with Brazil's case, may not achieve gender diversity as a goal.

However, these aggregate estimates showing the overall decrease in gender gaps mask important heterogeneity in the way discretion was reduced in practice. Exploiting another unique feature of my data detailing the specific screening methods for all job processes, in the second part of the paper I combine a theoretical framework with a multiple-treatment empirical strategy to isolate the effect of five alternative screening changes to reduce discretion in the sorting and selection of job seekers.

Before the impartiality requirement, occupations screened candidates using written, nonwritten, or a combination of both steps. The initial set of hiring tools in an occupation defined treatment assignment, a feature that I exploit for my empirical design and to test the identifying assumptions. Occupations with pre-treatment use of written tests fully blinded the exams; those using interviews or practical, oral exams either switched to a blind written test, or kept the non-written tools but added the less subjective step; those with a pre-existing combination either removed the non-written step or kept it. These different changes in screening methods, which define the treatments, all targeted a reduction in the level of discretion.

To validate the treatment assignment process induced by compliance with the new constitutional text, I conduct a series of tests. The space of possible treatments for an occupation includes at most one out of two changes in screening methods. This "one treatment or another" design where job processes are treated based on plausible exogenous variation generated by their job titles allows me to conduct three tests to support my empirical strategy. First, pretreatment covariates including degree of feminization in the occupation, and share of female applicants and competitiveness of the job process, collectively explain a negligible portion of treatment assignment.

A second test addresses the possibility that policymakers based the changes in screening for an occupation on treatment effects, so that relevant factors in their information set uncorrelated with the previous covariates could drive selection. But to the extent that these unobservables also matter for gender hiring gaps, they would plausibly lead to pre-treatment differences in the outcome paths of jobs that sorted into alternative treatments. This suggests a visual inspection that combines the usual pre-trends probe between treated and control with the evaluation of systematic differences in the outcome between occupations following alternative treatments. Gender gaps pre-treatment show parallel trajectories and indistinguishably-estimated differences in magnitudes between the pairs of occupations that could in theory follow alternative treatments. This indicates that, on average, selection of an occupation into a particular treatment shows no relationship with pre-existing differential gender hiring gaps, which in turn would be likely to pick up time-invariant unobservable factors. Finally, I leverage the fact that many occupations determine a treatment to assess systematic differences in estimated effects between each occupation, which further lends credibility to my design.

I build on a canonical model of statistical discrimination by introducing two new features. The first is tool discretion, which regulates evaluators' expression of their own bias. The second is tool bias, which allows for disparate impact independently of evaluator behavior. Written and non-written screening tools differ in the level of discretion, bias, and precision, closely matching the empirical setting. The model pins down gender hiring gaps in each of the five potential treatments that correspond to the different ways to reduce discretion as a function of three forces. These reduced-form predictions will guide the interpretation of the treatment effect estimates, many of which can have a theoretically ambiguous impact on diversity due to trade-offs between information an bias.

I break down the impact of each of the different ways to reduce discretion by first looking at the treatments that kept the tools in use. Starting with occupations that fully blinded job processes, the estimated impact on the gender hiring gap is a decrease of about 33% at the baseline. This treatment captures the causal effect of removing all sources of evaluator bias, and therefore reveals that it was an important force driving gender disparities.

The other treatment of this kind — occupations initially using a mix of written and nonwritten methods — kept the subjective stage and blinded the written step. Similar to fully blinding, this change in screening does not impose a potential trade-off because overall precision and bias remain unchanged. While the average treatment effect of partially blinding is null, the estimate is a function of the exogenously-determined weights given to blind written stages in job processes. In hiring processes with greater weight to fully blind exams, the gender hiring gap also considerably narrows. Importantly, I also show no evidence of strategic response by evaluators in non-written scores as their importance to the overall hiring decision varies across processes.

The second set of estimates comes from treatments that involved changes in the tools being used to screen job applicants. The first treatment introduces an objective tool to a preexisting interview. A common objection against requiring standardized tests is that they could disadvantage minorities through a disparate impact channel, if for example women are worse test-takers in a way that is not predictive of worker productivity.⁵ While my theoretical predictions show that this is a possibility, as long as these exams impose a smaller disparate impact than interviews, overall bias in screening will be lower. Combined with greater precision from their introduction, which favors minorities because performance signals influence evaluators' (men-favoring) priors more strongly, this condition would predict an increase of women hiring rates. Empirically, I estimate a decrease in the gender hiring gap of about 5.9 percentage points, or 35% of the initial hiring gap from using only an interview.

The next treatment removes subjective stages from the selection process. Similar to the other treatments that contain a mix of high and low-skill, here careers like cleaners, programmers, civil engineers, and nutritionists adopted the same change in screening. If evaluator bias matters and bias is particularly enabled by tools with high discretion, removing them from selection processes has an intuitive appeal. Contrary to this, my estimates reveal that removing subjective stages has no effect on the gender gap. This is consistent with a bias-information trade-off where the loss in screening precision from losing a productivity signal offsets the potential gains from eliminating disparate treatment and impact with the use of the tool. This results in evaluators putting more weight on their prior, making above-average women more difficult to be identified.⁶

To conclude the main set of results with multiple treatments, I also find that the way employers screen has important sorting effects. Changes in screening that improve perceived fairness without making the process more costly — because of additional hiring steps — increase women's application rates relative to men, further accounting for the increase in diversity captured by aggregate estimates. Taken together, I find that changes in screening tools have important consequences for both employer hiring decisions and job seekers' application behavior.

This paper makes several contributions. First, to the personnel economics of state literature (Finan et al. (2017), Besley et al. (2022)), which has studied governments' long-held goal of making civil service more professionalized (Grindle (2012)) in the context of large-scale reforms that curbed political patronage (Aneja and Xu (2022), Moreira and Pérez (2021a), Moreira and Pérez (2021b), Estrada (2019), Ornaghi (2019)). One limitation from prior studies stems from the overreaching essence of these reforms, where often the insulation of hiring from political influence also extends to protection from arbitrary firings, how promotion decisions are made, and general job incentives based on political cycles. This has made it difficult to isolate responses

⁵For example, Baldiga (2014) finds that women are more likely to skip than to guess on SAT questions that penalize a wrong answer, which decreases their test scores. Importantly, the pattern is not explained by gender differences in knowledge or confidence.

⁶The final treatment, where occupations change the screening design completely by replacing an interview with a written stage decreases the gender hiring gap by about 41%, another large estimated effect.

due to hiring design from changes in job attributes. My setting allows me to separately identify the role of discretion in hiring. Additionally, this is the first paper to directly observe and document how public servants are screened, overcoming another limitation of prior studies which could only provide little information into how governments carry out meritocratic selection due to data availability. By further observing the pool of applicants for the universe of government postings, I also show that different screening strategies employed by the public sector affect who applies and ultimately becomes a public servant, with a direct impact to the gender composition of the workforce.⁷

My findings also contribute to the literature on discrimination in hiring. Papers including Goldin and Rouse (2000), Bertrand and Mullainathan (2004), and Kline et al. (2022) have made great headway in documenting the existence of discrimination in hiring. But because of the nature of their evidence, prior work has been unable to identify whether disparities are driven by evaluators or the hiring practices used by employers. Combining the unique ability to observe the implementation of hiring practices with an empirical setting that isolates changes in screening methods, I show that hiring practices are a key source of gender gaps. Screening tools ultimately regulate the expression of bias, and directly targeting how screening is conducted can be an effective way to foster diversity.

This paper further provides insight to a strand of studies analyzing the importance of the degree of subjectivity of assessments in organizations. While much of the literature has focused on performance evaluations for workers once they are hired (e.g. Prendergast and Topel (1993), Prendergast and Topel (1996), MacLeod (2003), Frederiksen et al. (2020), De Janvry et al. (2023)), Autor and Scarborough (2008) and Hoffman et al. (2018) have analyzed the role of tests on firm hiring. This study augments this research by considering a broader set of choices employers have at hand to decrease subjectivity — remove discretion from tools already in use or replace, remove, or add tools with different levels of discretion. This provides important lessons: not all approaches to decrease discretion have the same implications for diversity, particularly when achieved through a removal of performance information about a candidate.⁸ I estimate that a reduction in bias at the expense of losing screening precision may be ineffective on average,

⁷This also contributes to a strand of work that focuses on which factors can affect the applicant pool to public sector positions, including with the use of financial and career incentives (Dal Bó et al. (2013), Deserranno (2019), Ashraf et al. (2020)). Relatedly, my results add hiring practices and their role in shaping anticipated discrimination to factors influencing the gender make-up of job applicants (e.g. Flory et al. (2015), Kuhn et al. (2020), Abraham et al. (2020), Card et al. (2021), Delfino (2021), Flory et al. (2021), Kuhn and Shen (2023)).

⁸Moreover, a growing literature demonstrates the potential negative consequences to diversity from partially concealing or removing non-performance information about job applicants, e.g., ethnicity (Behaghel et al. (2015)), age (Neumark (2021)), credit information (Bartik and Nelson (2022)), criminal records and history checks (Holzer et al. (2006), Agan and Starr (2018), Doleac and Hansen (2020)).

with evidence from additional analyses suggesting that this bias-information trade-off may vary depending on occupational skill.

Finally, this work provides a methodological contribution to the growing use of text analysis tools in empirical economics. Researchers have relied mostly on *ad hoc* dictionary methods to parse and interpret information in text form into a predictor of underlying phenomena (e.g., Gentzkow and Shapiro (2010), Baker et al. (2016)). More recent methods are useful in applications with structured layouts to identify text regions (Shen et al. (2021)). In many cases, however, researchers are interested in extracting actual structured data from text, a task that is especially challenging when the text is displayed without regular layout and contains confounding information. The natural language processing algorithm I develop leverages semantic patterns of raw text surrounding numeric data, without requiring structured layouts. This query-based approach offers a text analysis tool to enrich new methods being developed in economics.

The paper is organized as follows. Section 2 provides the institutional details and the main source of variation used in the paper. Section 3 describes the natural language processing algorithm I develop to distill hiring records into data. Section 4 shows the aggregate results of reducing discretion on sorting and selection. Section 5 sets up a model and empirical strategy to identify the effects on gender gaps from different changes in screening methods. Section 6 concludes.

2 Institutional Details and Setting

2.1 Hiring in the Brazilian Public Sector

Brazil has been using a formal, merit-based civil service hiring system for decades, being considered a primary example of a meritocratic and legally professionalized bureaucracy (Grindle (2012)). Since the 1960s, over 80% of public positions have been filled exclusively through this mandatory competitive examination system, known as *"Concurso Público"*.⁹

Public sector hiring follows strict rules and is organized in the following way. A number of constitutional requirements apply to all job selection processes at federal, state, and municipal levels, with each public employer implementing and conducting their own hiring. The first rule that all hiring processes must follow is transparency. Every step of the the *concurso* is pub-

⁹See Figure A.1 for a complete history of meritocracy implementation and public servant selection rules. There are strict limits to the number of openings and the types of jobs that can hire outside the *concurso* system. The remaining 20-30% public sector jobs legally exempt from formal civil service exams are temporary jobs, positions of trust, and commissioned posts. These posts are particularly common around politicians and specific bureaucrats, like congressional staff.

lished in a designated daily government gazette (similar to the Federal Register in the US). The second rule is that job openings must be allocated following the final ranking of candidates, which is exclusively determined by their scores in the screening steps.

Public hiring processes begin with a job announcement posting, called *Edital*. Job announcements need to detail the number of job openings, application requirements, and job attributes, and crucially for the research question, which screening methods will be used, and the weight of each exam toward a candidate's final score. Occupational attributes, including wages, are fixed by bureaucrats at high organizational levels and not subject to bargaining or employer discretion. The rules and details laid out in the job announcement are legally binding. In practice, this means that evaluators cannot ignore or modify a screening step.

The hiring process proceeds according to the general steps depicted in Figure 1. First, candidates apply to the job opening, have their applications screened based on announced requirements (e.g. attain the education level required), and have their names published on a subsequent journal issue. At this stage, the entire pool of candidates is publicly visible. The employer conducting the hiring process then publishes individual performance (scores) for each selection stage as the hiring process unfolds, including interviews and tests, identifies the candidates who are ultimately offered jobs, wait-listed, and hired. This concludes each *concurso*.

2.2 Impartiality Requirement

I exploit the introduction of a new requirement that altered one feature of public employee selection, while leaving the overall hiring procedures, rules, and the organization of public careers unchanged. The new requirement imposed that screening would have to be impartial, which in practice resulted in a shock to screening tools that reduced discretion.

In October 1988, Brazil passed a new Constitution expanding state guarantees and responsibilities as the country exited a 25-year period under military dictatorship. Known as "Citizen Constitution", the new legal framework modified several areas in the pre-existing constitutional text to align with the new social focus. This included the legal text organizing the hiring of federal public workers, which began to include a new provision mandating impartial screening.

The previous Constitution stated, regarding public sector hiring, that "*Public sector positions are accessible to all Brazilians* [...] *and hiring must be conducted through formal process (concurso) using exams or exams and candidate qualifications*" (1967 Constitution of Brazil, Section 7, Article 95). The 1988 Constitution amended this text by including a provision that the selection process should follow the principle of "impartiality" (*impessoalidade*). The new requirement aimed to codify the understanding that practices that could give an unfair advantage to an individual would violate the new Constitution and therefore be legally challenged. There were no specific directions or implementation prescription given, only that all hiring had to comply with the stated requirement.

2.2.1 Implications for the selection of public employees

The impartiality requirement generated a variety of changes to screening steps aimed at decreasing discretion. That is because, due to the lack of specific implementation rules in the Constitution, federal bureaucrats and government legal experts at high institutional levels provided different guidelines to public sector hiring. Keeping with tradition of organizing rules related to public servants around careers, discussions on new hiring practices were made at the occupation level. These centralized recommendations would then be implemented downstream by all federal employers.

Changes in screening methods followed two criteria. The first criterion focused on the legality aspect, which led to two universal changes. First, all occupations using written tests had to impose a blind implementation. Second, occupations could no longer screen candidates using exclusively non-written methods. The conclusion by government officials was that allowing for a candidate to be identified in a written exam could enable unfair treatment, and that relying only on subjective screening steps could lead to hiring decisions without a clear way to validate candidate performance (lack of "receipts").¹⁰

The second criterion was that, given that modifications were deemed necessary, those were made striking a balance between compliance with the impartiality requirement and maintaining screening tools considered important in a particular job. These considerations have been widely cited in Brazilian courts, which confirms their relevance in how treatment assignment was determined. For instance, in a 1993 legal proceeding (TC 013.813/93-5) involving a hiring process for journalists by the Federal University of Pernambuco, the Federal Court of Accounts (TCU) stated that a practical exam in that specific occupation was desirable, since "journalists are commonly on camera and therefore observing their identity is necessary".

¹⁰Figure A.2 shows an example of the language that began being added to job processes post-1988 about blind exams. In this selection process for federal judges published on September 4, 1989, a rule states that candidates identifying themselves in written exams will be excluded from the hiring process. This language was virtually nonexistent in announcements before the introduction of the impartiality requirement. In line with that, the website Jusbrasil.com, which compiles over 100 million of litigation cases in Brazil, lists no civil proceedings involving applicants removed from job processes because they identified themselves in the exam until 1988, while after the new Constitution there are hundreds of cases related to non-compliance with blind exams.

2.2.2 Treatment timing and assignment

Preexisting screening methods in an occupation therefore drove which screening changes were adopted to comply with the impartiality requirement and reduce subjectivity, providing exogenous variation in screening tools. Empirically, 95% of hiring processes within an occupation in the federal sector modified screening in the same way, underscoring the top-down nature of treatment assignment, which was not determined by employers or evaluators conducting hiring. Each change in screening methods involved a variety of occupations across several dimensions. For example, kitchen assistants and civil engineers had screening modified in the same way, just like teachers and police officers were screened the same way.¹¹

Important for my identification purposes, the introduction of the impartiality requirement did not attend specific policy goals. For example, if policymakers intended to increase the participation of minorities in government, the provision could have been added with the intent to reform hiring. If that translated into an active implementation, employers or occupations deemed more problematic (e.g. lower women representation, more intensive use of discretionary screening tools) may also have been treated differently. In contrast with these considerations, evidence from over 14,000 pages of the Constituent Assembly records that I investigate show a clear pattern of only procedural or technical discussions around the implementation of the new provision.¹²

The provisions in the new Constitution applied to public sector hiring at all government levels. However, because public servant selection is conducted by states and municipalities independently of the central authority, states had to individually pass the appropriate legal framework to comply with the new impartiality requirement. It took several years for the sharp shift in federal employer behavior with respect to hiring to trickle down to state agencies and governments. This means that, while state jobs are suitable as the control group around the Constitution introduction, this institutional feature also limits the duration for which I can credibly study post-1988 outcomes. Since the first changes in screening methods in state job announcements start to appear in 1991, I limit the baseline analysis in the paper until 1990.¹³

¹¹See Table A.1 for several examples of occupations in each treatment.

¹²Constituent Assembly discussion records are retrievable, in Portuguese, here https://www25.senado.leg.br/web/atividade/anais/constituintes#1988 and here https://www.senado.leg.br/publ icacoes/anais/constituinte/sistema.pdf.

¹³Due to the lack of systematic roll-out across states, many of which officially passed the requirement much later, but already showed informal changes to screening gradually over time starting in the early 1990s, alternative empirical strategies exploiting time variation in adoption across federal and states are not well-suited.

2.2.3 Broader institutional context around the new Constitution

While the impartiality requirement effectively represented a shock to public sector hiring, its introduction occurred with Brazil's new Constitution, which bundles broader social and institutional changes, including the exit from a period under military dictatorship. This potentially introduces contemporaneous changes that could confound the analysis. Importantly, my empirical design is only subject to identification threats that shift unobserved attributes of federal jobs relative to state jobs. This could be the case if, for example, the Constitution established female-friendly job amenities for federal positions only. In contrast, factors like the expansion of services provided by the federal government or modifications in the tax code would not be consequential.

Alleviating this concern, there were no relative changes to labor markets between federal or state governments or even between public and private sectors more broadly. Institutional changes introduced around the same time like the shortening of the workweek or extension of maternity leave were rights extended to all formal jobs in the economy.

3 Opening the Hiring Black Box

3.1 Raw Text Sources

The raw data used in this paper come from over 35 million of official government journal pages from Brazil (known as "*Diário Oficial*") since 1980. These gazettes are similar to the Federal Register in the US and publish the universe of public notices spanning public procurement processes, executive orders, and information on public servants, including hiring. Every government branch maintains its own decentralized repository with daily scanned issues of official journals. Many journals require a page-by-page access point.¹⁴

3.2 A New Approach to Transform Unstructured Text Into Data

3.2.1 Practical challenges

While these documents provide a rich source of hiring records, two main challenges prevent the direct retrieval of data on selection processes from Brazil's government gazettes.

First, there is no systematic way to search, identify, or link job hiring processes in the government journals. Hiring processes are mixed-in with extraneous and highly confounding

¹⁴Table A.2 shows a complete list of the separate government entities used to retrieve the gazettes, as well as when issues first become available online.

information, and postings from the same process published over time do not have an identifier that enables linkage. Information confoundness is a key issue which makes off-the-shelf text analysis tools based on similarity or proportionality ineffective.¹⁵

The second challenge is that, even if job processes could be identified and linked, text content in the government journals is highly irregular and unstructured. This is because there is no pre-determined layout that public agencies must follow when publishing notices, which results in an unwieldy variety of "shapes" to be handled by conventional scraping targeting ex-ante known structure patterns (e.g. a table with certain layout or free text organized in two columns).¹⁶ As a consequence, an alternative approach to extract information not based on layout-pattern recognition is necessary.

3.2.2 Two-step natural language processing algorithm: Summary

To address all of these challenges, I develop a two-step natural language processing algorithm that allows me to first define the relevant text portions from highly confounding text, attribute a posting to a unique job hiring process, link its different postings, and finally transform unstructured text into data. The algorithm generalizes a search query with learning and can be applied to a wide variety of empirical settings that follow the same general extraction data problem and goal of this paper.

Because the algorithm's conceptualization and progression involve an extensive number of rules and technical considerations, I detail its implementation in the Appendix B. To fix ideas here, I use Figure 2 which summarizes the implementation mechanics and data output of the algorithm using one job process as example.

Figure 2 shows screen captures of the raw government gazette pages containing the (a) announcement, (b) list of applicants, and (c) final results for a job process in the federal government to select public prosecutors. For visualization purposes, the announcement and list of applicant posts trimmed and additional posts containing exam scores before the final results are not shown. Text snippets bounded in red are extraneous posts unrelated to the target hiring process, while text bounded in green represents the correct contents. Note that the target text on the first panel is organized in paragraphs within numbered sections, on the second in

¹⁵Standard text analysis algorithms that are increasingly popular in economics (like term frequency-inverse document frequency and cosine similarity) are poor tools for connecting different text corpora based off *exact* text vectors. Even the sophisticated lexical fingerprinting tools used to detect plagiarism would still rely on the resemblance between text documents that might not be informative for linking purposes. These algorithms require calibration that is context-specific, demanding supervision in a large number of cases, drastically decreasing gains to automation and resulting in a large number of type I and II errors.

¹⁶See Figure A.3 for some examples. I document over 200 different text layouts, with multiple variations within the same broad layout type.

continuous-like text, where applicant names are divided by a delimiter (;), and in tables in the last panel.

There are two important ideas behind the architecture of the algorithm's first step: posts in the government journals are temporally ordered — a job announcement must be published before its hiring decision. This means that when searching for a complete job process, the algorithm first identifies an announcement, then proceeds to journals published after the announcement date as it searches for the other posts.

Second, because these are official records subject to legal enforcement, each post must have enough information to enable a human reader to distinguish one job process from another. How much information a computer needs to perform entity linking as efficiently as humans is a complicated and actively-discussed problem, but I consider that there must exist a sufficient set of attributes comprised by a message keyword (job), sender (establishment), and keyword feature (which may be needed to resolve ambiguity). In Figure 2, the message keyword takes the correspondence *public prosecutor* and the sender is the *Office of the Federal District Attorney General*. Because that establishment did not have any other job hiring processes for the same job title ongoing until the final result is detected, establishment and job are sufficient to link the different posts of this hiring process.

Having determined that the three posts shown in Figure 2 belong to the same hiring process, the next step is to extract the data. For simplicity, I illustrate this step focusing on the shaded green area in panel (c), which includes the final scores for all job applicants who made to the end of selection. In this step, the algorithm exploits the semantic relationship within a text snippet (instead of across) leveraging the fact that, if I am interested in retrieving numeric strings (the scores), each number can be seen as a target of a message. Thus, every score must have a sufficient number of attributes: a sender (the evaluator), a receiver (job applicant), and a feature (exam type). This approach is flexible because it does not impose a physical relationship between these textual elements that will be carried to the data output; rather, it assumes that for a human to properly assign meaning to a number, these attributes must be in the neighborhood of that number.

3.2.3 Data validation and estimating sample

I organize the algorithm's pipeline that implements step 2 following the sequence in Figure A.4. Implementation performance can be thought as an inference problem: all text snippets extracted in the first step contain the true data, but also false positives (incorrect signals). The goal is to separate the two. To ensure the final dataset contain only correct information, the algorithm first maximizes stringency and therefore the number of false negatives — it throws

away important information that should be part of the output. Then, to recover discarded information, I increasingly relax the search parameter's stringency.

In a world without computation constraints, this iterative process would continue until gains in data extracted between two iterations is negligible. At a given point in the pipeline, my data dimensionality spans anywhere from 8 to 60 billion characters, implying more than 2 trillion matching computations needed every time the NLP algorithm runs. To make extraction feasible, I continue to minimize false negatives in the data output until the final data has an accuracy above 90% and the cost of decreasing false negatives is greater than some threshold increase in the number of observations in the final data (e.g., a manual adjustment to the algorithm that takes 1 hour produces 5 additional job processes that were otherwise discarded).

While there are no official sources to validate the data extracted with my approach around the impartiality requirement introduction, later periods have external aggregate statistics that I can use to run evaluation exercises. Figure A.6 plots annual official records provided by Brazil's federal government starting in 1995 containing the number of selection processes, number of hires, share of female hires, and share of high-skill jobs. Next to each statistic, I construct the equivalent measure based on data generated by the algorithm. Though this is out-of-sample in the time-series, the raw source and overall quality and layout of documents are the same as in the 1980s and 1990s, therefore likely to provide a reliable benchmark. The correlation between the generated data and the official records is on average 99% for each series, with point error incidence lower than 2%.

For the analysis in this paper, the sample includes the federal government and the states with official gazette issues available online for the period. These states were Amazonas in the country's north region, Pernambuco in the northeast, Distrito Federal, Mato Grosso, and Mato Grosso do Sul in the central region, São Paulo — the largest and richest state — in the southeast, and Rio Grande do Sul in the south. I use all job processes with complete information on job requirements, screening steps, as well as candidate scores, final ranks, and job offers, if any.¹⁷ I focus the analysis on the 1986-1990 period since states began jointly passing new state Constitutions with similar guidelines to the federal rules in the early 1990s. In the case of states, however, the enforcement of impartiality rules was much less organized, with some state employers changing hiring methods in the 1990s and others in the same state delaying implementation, which for example invalidates a staggered implementation design.

Figure A.5 shows the gender distribution of applicants in the resulting sample by occupation and skill level, and Table 1 provides summary statistics on education requirements,

¹⁷There are no reasons to expect systematic factors explaining why some states lack online archives of government gazettes available in the 1980s or why states in the central region is over-represented. Mato Grosso and Mato Grosso do Sul share the same digital archive provider and software, and the Distrito Federal (akin to Washington DC in the US) is the seat of the federal government.

screening steps, and job applicants for control and treated groups, before and after the reform. There is considerable gender sorting across occupations and characteristics of posted jobs also vary over time. These composition effects motivate the inclusion of job fixed effects in all my specifications.

4 Overall Effects of Decreasing Discretion in Hiring

This section shows the aggregate results on gender sorting and selection from a reduction in discretion in hiring. The unique ability to observe how public servants are screened, evaluation and outcomes from screening, and the applicant pool, enables the analysis to proceed in the following sequence. First, I verify federal jobs' compliance with the new requirement, also confirming that state jobs did not modify their screening around treatment. Then, by disentangling supply and demand I show that greater impartiality in hiring improved women's final scores relative to those of men, driven by higher marks in objective exams. This translated into a higher probability of hiring. Finally, I leverage the hundreds of occupations, across all skill levels, in the data to check how these results interact with different job characteristics of public servants.

4.1 Did the Impartiality Requirement Change Screening Tools?

I begin by investigating whether and how federal government hiring processes complied with the impartiality requirement. Specifically, I compare how an occupation hiring in the federal sector modifies screening in a variety of ways consistent with decreasing subjectivity and discretion, relative to the same occupation hiring in states.

Figure 3 compares average changes in screening methods for state jobs to changes in federal jobs. The statistics plotted are averages across within-occupation averages, which means that they are free of compositional changes in the supply of different jobs. Comparing similar occupations between treated and control groups is important because their relative make-up may be unbalanced (e.g. healthcare services are usually provisioned at the local and state levels) and fluctuate over time due to different shocks to state and federal governments, including political cycles and budget decisions.

As expected given the institutional context of the new Constitution, state jobs show no response in the use of written and non-written screening tools, which is also reflected in the same number of screening rounds implemented. This is also consistent with a lack of strategic response by state employers, underscoring the rigidity of how screening steps are defined in public sector processes. In contrast, federal jobs adjust sharply by decreasing subjectivity, both

by reducing the proportion of non-written tests and increasing reliance on written tools. Because these variables capture a compliance margin without blinding effects, these results show that federal job processes also actively modified screening steps. This led to almost double the number of hiring stages on average, driven by the inclusion of blind written tests to the existing process.

4.2 Empirical Strategy

My baseline empirical strategy implements a triple-difference specification comparing changes in outcomes in the same occupation for federal and state jobs. Specifically, I run:

$$y_{ijt} = \delta_{1} \text{Female}_{i} + \delta_{2} \text{Federal}_{j} + \delta_{3} \left(\text{Post}_{j(t)} \times \text{Federal}_{j} \right) + \delta_{4} \left(\text{Post}_{j(t)} \times \text{Female}_{i} \right) + \delta_{5} \left(\text{Federal}_{j} \times \text{Female}_{i} \right) + \beta \left(\text{Federal}_{j} \times \text{Post}_{j(t)} \times \text{Female}_{i} \right) + \delta_{j(o)} + \theta_{t} + u_{ijt}$$
(1)

which tracks outcomes for applicant *i* in job process *j* in a given year *t*. If a job process occurred $t \ge 1989$, Post_{*j*(*t*)} is a dummy equal to 1, and if it was for a federal job, Federal_{*j*} = 1.¹⁸ The dummy Female_{*i*} corresponds to whether the job applicant is a woman, and the specification controls two-way for occupation and year fixed effects. The triple-difference estimate β measures how gender gaps in the federal sector changed after the decrease in discretion compared to changes in gender gaps in the same occupation in state jobs. Relative differences between the same occupation in federal or state employers, including posted wages or prestige, are differenced out by $\delta_{j(o)}$. Year fixed effects θ_t also control for aggregate time-varying factors like inflation shocks and spending cycles.¹⁹

In all specifications, standard errors are clustered at the job process level. Because scores and offers are made within a selection process, this level of clustering allows for correlated factors across job applicants: a job process with greater candidate demand, a more stringent set of evaluators, exam contents, and so forth. I assign gender to job applicants using Brazil's Census Bureau *Gender of Names* database, which contains nearly 200,000 unique first names and their corresponding gender. The matching precision between job applicant names and gender from this dictionary is above 98%.

¹⁸Since the new Constitution came into effect in October 1988, I consider the post-treatment period as the beginning of 1989. In line with that, I start observing changes in hiring processes announcements in 1989.

¹⁹Specification (1) contains the additional δ_1 and δ_5 parameters since the unit of observation (job applicant *i*) is not nested within the treatment level (federal government). This generates variation to both parameters beyond the fixed effect δ_2 . The interpretation of β remains the usual triple-difference treatment effect.

Assessing the plausibility of the identification strategy. The main identification assumption is that, in the absence of the institutional change, gender gaps in federal and state job processes would exhibit parallel trends. Figure 4 shows event-study style estimates of model (1) separately for treatment and control groups. The panel on the left plots gender application gaps and the one on the right gender score gaps. This visualization helps to conduct the usual pre-trend assessment, and also to inspect the behavior of gender gaps separately for federal and state jobs around the introduction of the new Constitution.

Pre-treatment paths for gender hiring gaps are parallel between treated and control groups, indicating an absence of pre-trends. Following treatment, the outcome in the federal government shows a sharp change as soon as new job processes under the new requirements start to roll out, while the gender hiring gap in state jobs remains with almost an identical magnitude relative to pre-1989. State employers show no sign of "front-running" the implementation of impartial screening by their legislative bodies, attempting to hire more men before the requirement came into effect outside my estimation window.

A potential concern in difference-in-differences designs is the violation of the Stable Unit Treatment Value Assumption (SUTVA) assumption. In my setting, the shift in how screening had to be conducted in the federal sector could have changed application rates to state jobs in ways inconsistent with this assumption. One example would include the possibility that women start applying more often to federal jobs in detriment of state jobs or that men directed job search to states, avoiding a decrease in discretion in the treated job processes. Panel (a) of Figure 4 shows that gender application gaps in state hiring processes remain stable over time, overlapping very closely with the outcome in federal jobs pre-treatment. This indicates, in addition to a lack of pre-trends again, no evidence of spillovers to control job processes.

4.3 Disentangling Hiring Supply and Demand

Observational data available to researchers often allow one to only observe hires as the outcome of a selection process. However, changes in screening can affect both employers' hiring decisions, as well as job seekers' application decisions. In order to separate between these demand and supply effects, one necessarily requires knowledge on the applicant pool, which is a unique feature of my data of public sector labor markets in Brazil. To see this more clearly, consider the following decomposition:

$$\underbrace{\mathbb{P}(\text{Female}|\text{Hired}=1)}_{\text{Share of female hires}} = \underbrace{\mathbb{P}(\text{Hired}|\text{Female}=1)}_{\text{Demand}} \times \underbrace{\mathbb{P}(\text{Female}=1)}_{\text{Supply}} \times \underbrace{\frac{1}{\mathbb{P}(\text{Hired}=1)}}_{\text{Competitiveness}}$$
(2)

which separates the choice of women to participate in the hiring process from the conditional probability of them being hired. Once adjusted by the competitiveness of the job process — the probability that an applicant of any gender is hired — they jointly determine the share of female hires. Systematic gender differences in the probability of who applies and who gets selected therefore determine gender gaps in hires.

While the share of female hires captures whether the workforce is becoming more diverse, separately estimating selection (demand) and sorting (supply) effects is important to understand how each margin adjusts to changes in the hiring process design, since they have distinct potential remedies and policy consequences. For example, suppose that the share of female hires at an establishment is low. This implies (i) a potential propensity of the employer to discriminate (under a set of assumptions) against women, (ii) lower quality or fewer qualified women applying for the job, or both. As a consequence, using $\mathbb{P}(\text{Female}|Hired = 1)$ amounts to conditioning on an endogenous variable (the available candidate pool), which can lead to incorrect inference about the existence of employer bias. This could result in the inaccurate enforcement of anti-discrimination laws. It also provides employers with little information about the sources of the observed lack of female representation and how to best address it.

On the demand side, biased screening can over-select men by making them seemingly better than equally qualified women. But the use of biased screening can also have a gendered effect on supply, leading to suboptimal hiring if more qualified minority candidates refrain from applying. That is because women may react to even perceived unfair treatment (Small and Pager (2020)).²⁰ That further interacts with gender differences in behavior involving competitive environments, which includes job selection processes.²¹

Exploiting the ability to separate these two important channels, I organize the results below on the overall effects of greater impartiality in hiring in the following sequence. First, I show how the new requirement affected gender score gaps, which leverage detailed records for applicant performance in each screening stage of a job process. Then, I measure changes in gender hiring gaps, i.e. the labor demand component $\mathbb{P}(Hired | \text{Female} = 1)$. Finally, I estimate gender gaps in supply responses ($\mathbb{P}(\text{Female} = 1)$), ultimately assessing the importance of each channel to affect employee diversity.

²⁰Women are more likely to place greater weight than men on fair treatment, and the perception of fair treatment is more strongly linked to women's than to men's willingness to apply at a previously rejecting firm (Brands and Fernandez-Mateo (2017)). There is also evidence that women respond to nudges that they are "welcomed", as with ads stating a preference for diversity (Flory et al. (2021)).

²¹Extant literature shows that women are less likely to apply for promotions and less likely to enter tournaments due to a lower willingness to compete or self-stereotyping (Niederle and Vesterlund (2007)), Hospido et al. (2019), Bosquet et al. (2019), Coffman et al. (2023)), and tend to sort into female environments to avoid competing against men (Gneezy et al. (2003)).

4.3.1 Effects on gender score gaps

Table 2 begins by comparing final scores in the hiring process received by female and male candidates. Scores are standardized within each job process, so that they are comparable across all processes. Final scores of women increase by 0.07 standard deviation after the new Constitution is implemented, and the final scores of men decrease by slightly more. Combined, these effects imply a 0.14 standard deviation reduction in the gender score gap, which closed the pre-existing gap.

Next, I break down scores by screening tool type. Columns (4) through (6) show that written exam scores, which necessarily became blind for federal jobs, improve significantly for women's relative to men. The overall effect on the gender gap in written scores is about 0.13 standard deviation. Columns (7) through (9) repeat the analysis for scores in non-written exams, which, unlike written tests, were not directly altered in response to the impartiality requirement. Results show no changes in gender gaps for oral, interview, or practical exam scores, confirming that evaluators did not seem to adjust non-written scores strategically in response to treatment.²²

4.3.2 Effects on hiring decisions

Next, I investigate how the closing of the gender score gap translates into hiring odds. Candidates' final scores determine their ranking in the job process, with the highest-scoring applicants being hired.

While the scores of the average woman applicant increased relative to men, this may not necessarily imply a narrowing of the gender hiring gap. For that to happen, women who were just not hired must be able to take over the position of the marginally hired men. If overall increases in scoring came from the highest or lowest performing women their hiring status would likely not change. Table 3 confirms that the higher female performance translated into improved hiring odds for female applicants. Columns (1) and (2) show that the probability of being hired for women increase while for men it decreases. Women become 0.3 percentage point more likely to be hired after the reform and men's hiring rates decrease by about the same magnitude.

Combined, the point-estimates imply a reduction of the gender hiring gap of 0.7 percentage point (column (3)), equivalent to 44% of the initial hiring gap. The fact that the gender score gap closed but the hiring gap only narrowed, albeit considerably, points to increased perfor-

²²Note that the separate estimates on gender score gaps by screening tool type restrict the underlying sample to those hiring processes that had the tool before and after treatment. Thus, it excludes job processes that for example switched from a non-written to a written test, or that had a combination of both methods pre-treatment but dropped the interview. The analysis in Section 5 considers each specific change to screening methods.

mance effects somewhat dispersed across women's distribution. Raising scores for women at the left tail of the distribution is not enough to make them competitive, just like higher scores for those at the top who would already be selected do not convert to higher hiring odds. This suggests that the benefits of reduced discretion were not concentrated to applicants of only a certain quality.

4.3.3 Effects on application behavior

Next, column (4) in Table 3 estimates the supply response of female applicants, captured by the sorting term $\mathbb{P}(\text{Female} = 1)$. Application rates of women grew by 1 percentage point, for a baseline application rate slightly tilted toward female applicants in federal jobs. This shows that women application rates were relatively more responsive to greater impartiality in screening.

To gain further insight into the general forces behind supply movements, in Table A.3, I run job process-level versions of model (1), first with a focus on the number of job applicants. Estimates show some evidence of an overall decrease in the number of applications, although a statistically insignificant effect. This may be driven by an overall increase in the cost of participating in the average selection process, since there are more screening rounds after treatment. I formally investigate this possibility in Section 5, where I distinguish sorting responses to treatments that clearly reduced discretion without modifying the number and type of tools being used from the ones that did so. Results are consistent with women reacting to perceived fairer screening by increasing application rates.²³

4.4 Additional Results and Robustness Checks

4.4.1 Job applicant quality

The prior analysis does not directly control for potential changes over time in the composition of job applicants, since it relies on a repeated cross-section from the perspective of candidates. Because of that, the quality of applicants could vary around the treatment. If better-quality women began to apply for federal jobs at higher rates than men, at least some of the decrease in gender gaps could be attributable to changes in quality.

While this potential effect could still be due to a decrease in impartiality — perhaps better prepared women react more strongly to fairer screening since their outside option may be more

²³Another potential driver is the job process competitiveness, which tends to correlate with the size of the candidate pool, and is determined by budgetary and personnel management constraints on the supply of job openings. However, controlling for process competitiveness or announced job openings does not change these results.

valuable — it is an informative exercise to assess how much changes in quality matter for the results. To test for that, I include a job applicant fixed effect to the previous set of estimates, effectively controlling for changes in the distribution of any individual time-invariant characteristic. This strategy is enabled by two characteristics of my data. First, applicants' names and identifying information allow me to link them across *concursos* at any government establishment. Second, due to the highly competitive nature of selection processes, many individuals apply more than once.

Table A.4 replicates the results in Table 2 for final scores adding applicant fixed effects. The estimate for the increase in women's scores using the sample of repeat-applicants is larger than in the previous specification that allows for compositional changes, while the effects for men are more negative. In addition to confirming the baseline results, this specification further suggests that the average quality of the female applicant pool decreased, which is consistent with the response anticipated when the artificially high hiring threshold to select women pretreatment is lowered. In summary, the decrease in the gender score and hiring gaps were not driven by more qualified women applying.²⁴

4.4.2 Alternative empirical specifications

Table A.5 reports a series of alternative specifications of model (1). To assess the robustness of the baseline specification, I modify how standard errors are clustered, the set of fixed effects, controls, and how job applicants' results in the selection processes are measured. The estimated magnitude of the decrease in gender gaps is stable across all tests, which include clustering at employer, government, or occupation levels, and controlling for occupation skill, geographic region of jobs, interacted fixed effects, or applicant pool size.

4.4.3 Occupational differences in skills and female representation

The magnitude of the estimated treatment effects may depend on characteristics related to the occupation. Women may face more stereotypes for higher-skill jobs at the top of the career ladder, certain jobs main provide more easily observable productivity measures during screening, diminishing reliance on statistical discrimination, and thinner markets in highly-specialized high-skill jobs make taste-based discrimination more costly.

Table A.6 examines how the effect of increasing screening impartiality varies across the occupational skill level. Compared to male candidates in high-skilled occupations (college

²⁴This interpretation imposes the assumption that time-varying unobservables correlated with quality did not change differentially by gender over time. That is, female repeat-applicants did not become better relative to men repeat-applicants.

degree or more), female scores increase by 0.2 standard deviation, and their hiring rates by 1.1 percentage points. This represents a magnitude 50% greater than the average effect estimated across all skill levels. Increasing impartiality for low-skill jobs — that require at most a high school diploma — does not have an effect on scores or hiring rates of women.²⁵

Table A.7 assesses how my baseline estimates differ by the feminization rate of the occupation, assigning a job title to one of the three groups: female-dominated, neutral, or maledominated. Comparing columns (A1) and (A3) reveals that the final score gap narrows by approximately 0.3 standard deviations for both female- and male-dominated occupations. Investigating the effects separately by gender in panels (B) and (C) reveals that women's scores increase more in male-dominated occupations, while men's performance falls the most in feminized jobs.²⁶

The overall decrease in discretion closed the gender gap in hiring by almost half and increased female representation in Brazil's public sector. This was driven by both an effect on both employers' hiring decisions, as well as job seekers' application decisions with a higher share of women induced to apply. In the next section, I investigate which of the changes in screening adopted to reduce discretion in hiring account for these results.

5 Implications of Different Strategies to Reduce Discretion

The analysis so far shows that application and hiring rates of women increased in response to the decrease in discretion in job processes for federal employees. These aggregate estimates, however, mask heterogeneity in how screening methods were modified to reduce discretion. Distinguishing between these different strategies is important, since alternative ways to reduce discretion can have distinct consequences for diversity. The choice of screening tools may impose a bias-information trade-off: removal of subjective steps may reduce bias but also discards

²⁵One possibility accounting for this result is that the specific way to reduce discretion varied along the skill distribution. Another, that the same change produces different effects depending on the skill level. For example, screening tools that give evaluators more discretion may be more valuable in higher-skill settings as private signals observed by evaluators may be more important to determine worker quality relative to lower-skill settings. I investigate these points more formally in the next section.

²⁶I measure the degree of feminization as the gender segregation of the occupation in the public and private (when applicable) sectors, the total share of women applying to job openings of that occupation, and when the occupation requires a specific college degree (e.g., structural engineering), the national gender make-up in the major. These factors broadly align. Occupations with less than 40% female participation are defined as male-dominated, 40-60% neutral, and above 60% female-dominated. The overall response here is consistent with the previous results across all occupations, where the downward adjustment in men's scores is quantitatively larger than the increase for women's, $\left(|\hat{\beta}^F [\text{Female} = 0]| - |\hat{\beta}^M [\text{Female} = 0]|\right) - \left(|\hat{\beta}^M [\text{Female} = 1]| - |\hat{\beta}^F [\text{Female} = 1]|\right) \approx 0.04$. These results can be seen as the "benefit" men derived in female-dominated occupations being larger than the penalty women faced in male-dominated sectors. Similar results carry for gender hiring gaps.

information; while adding objective tools constrain evaluator behavior and bias expression, but can itself introduce disparate impact.

To estimate the equity implications of different screening strategies, I take advantage of two main features of my setting. First, the unique ability to observe the screening methods used in each hiring process, rarely disclosed by employers — the hiring black box. Second, pre-existing differences in how occupations screened before generates exogenous variation in screening changes as a response to the reform, which capture different ways to reduce discretion. Some occupations made the implementation of tools already in use more impartial by partially or fully blinding screening. Others changed the combination of tools being used: either by removing or replacing exams with high discretion (like interviews), or introducing a tool with no discretion (blind written tests).

5.1 Treatments Generated by Discretion Reduction

Screening methods used in job selection processes belong to one of two categories: *written* (*w*) and *non-written* (*nw*). *Written* tools include open-ended and sometimes multiple-choice exams. *Non-written* exams comprise oral and practical examinations, and interviews. Screening may use either written or non-written exams, or a combination of both. This results in varying degrees of discretion and subjectivity across occupations, which is the level of assignment of screening methods in Brazil's public sector.

Before treatment, an occupation uses one of the following three possible screening tools: $pre \in \{w, nw, w + nw\}$. With the impartiality requirement, the exclusive use of nw is discontinued and written exams become blind. This implies that there are two possible combinations of exams post-treatment: $post \in \{w(b), w(b) + nw\}$. I define a treatment d as a mapping from *pre* to *post*, that is, a change in screening methods. This approach is consistent with the institutional setting, since the pre-existing mix of screening tools in an occupation determines which post-treatment types of exams will be used. This one-to-one mapping between an occupation and the set of screening tools used reflects the fact that 95% of job processes within an occupation follow only one treatment d.



Specifically, occupations using *w* before 1989 can only blind fully blind the tool already in use, so that $d = w \longrightarrow w(b)$. Following the same reasoning, those occupations with *nw* pretreatment would need to introduce a blind test, generating one of two potential treatments: $nw \longrightarrow w(b)$ or $nw \longrightarrow w(b) + nw$. Finally, occupations with historical use of w + nw can be assigned $w + nw \longrightarrow w(b) + nw$ or $w + nw \longrightarrow w(b)$.²⁷

In total, the impartiality requirement generates 5 possible treatments, with each occupation being assigned one type of change of screening tools. Though all of these treatments intended to reduce overall discretion in job processes, the implication of alternative treatments to gender gaps at face value is unclear. That is because different design choices modify bias and precision during selection. For example, how to evaluate whether we should expect $nw \rightarrow w(b)$ to increase female representation more than $nw \rightarrow w(b) + nw$? The first treatment trades screening and bias from one tool for another: even if interviews allow for more evaluator bias to be expressed, written tests could impose a greater disparate impact. The second alternative treatment increases overall screening precision by adding a blind test, but could also result in a countervailing force due to added tool bias. To sharply answer questions of this kind, one needs a framework that delivers predictions for gender equity that are empirically relevant.

5.2 **Theoretical Framework**

While targeting an overall decrease in discretion, different modifications to hiring methods have unclear consequences for gender hiring gaps. That is because most potential treatments generated by the impartiality requirement may introduce a trade-off between bias and infor-

²⁷Figure A.6 traces out the complete potential treatment space and gradually eliminates those ruled-out by my empirical design or the underlying setting of public sector hiring. Cases shaded in gray in the space grid are ruled out in a sharp difference-in-differences design (perfect compliance). Of the six remaining transition cases, $w \rightarrow w(b) + nw$ accounts for less than 1% of transitions in the data.

mation. This subsection sets up a simple theoretical framework that generates empirical predictions on how each change in screening tools impacts hiring rates of men and women.

5.2.1 Model summary

I build on the canonical model of statistical discrimination with fixed wages (Phelps (1972), Autor and Scarborough (2008)), consistent with the setting in Brazil's public sector. My model introduces two new features: tool discretion, which regulates evaluators' expression of their own bias, and tool bias, which allows for disparate impact independently of evaluator behavior.

The first feature of my model is tool discretion, which captures the fact that different tools may allow for different degrees of evaluator bias expression. Evaluators have the task of selecting employees with a mix of screening tools delegated to them by the employer. I allow managers to have a systematic bias for a certain demographic group. The term could be interpreted as taste-based discrimination, or any other type of systematic bias, such as rewarding social skills that do not predict productivity, but favor a certain group. However, the expression of evaluator bias is regulated by how much discretion a specific screening tool enables. Interviews allow for high levels of discretion due to their subjective nature, while the results from written tests are more easily observable to the employer, making bias expression more costly. In contrast, statistical discrimination expression in the model is independent from the degree of discretion of screening practices used. Managers base their prior of a candidate's productivity on her group membership. When candidate identity is concealed, they instead resort to the population mean. Evaluator bias and statistical discrimination combine to represent disparate treatment.

The second addition I make is to allow screening tools themselves to be biased. Independently of evaluator preferences, certain screening tools may disadvantage a particular group. For example, if written tests reward risky behavior by penalizing wrong answers without measuring productivity, women may be disadvantaged and the screening practice would lead to disparate impact. The role of tool bias is equivalent to adding systematic noise to the productivity signal provided to evaluators, favoring less productive applicants of the favored group.

The model pins down gender hiring gaps in each of the five potential treatments that correspond to the different ways to reduce discretion as a function of evaluator bias, tool bias and precision. These reduced-form predictions will guide the interpretation of the treatment effect estimates later in the section.

5.2.2 Environment

An employer (the principal) delegates the screening of a pool of job applicants to an evaluator. The candidate pool comprises individuals from two demographic groups, $x = \{m, f\}$, corresponding to a minority and majority group, female and male, respectively. The employer bases the hiring decision on some indicators of productivity $\theta = \{s, \eta\}$, observable only by hiring evaluators, which noisily measure a candidate's true productivity level, y. The productivity of job candidates is distributed as: $Y \sim N(\mu_0(x), 1/h_0)$, where the mean $\mu_0(x)$ is allowed to depend on group membership, and h_0 is assumed to be independent of x. I assume women to be the minority group, in the sense that women's average productivity is perceived to be lower than men's, $\mu_0(f) < \mu_0(m)$.²⁸

The employer's objective is to hire a proportion *K* of workers that maximize expected productivity.²⁹ But the evaluator objective is imperfectly aligned with that of the firm. The evaluator cares both about productivity and her bias toward a group ($\pi(x)$), which must be jointly maximized when hiring job applicants by

$$u(y, \pi(x)) = y + (1 - c_{\theta})\pi(x) \equiv y + d_{\theta}\pi(x)$$

where c_{θ} is a cost or penalty function defined over $c \in [0, 1]$. This component captures the cost that the evaluator faces by expressing bias, i.e., reporting to the employer a value of a candidate's measured performance that differs from the signal provided by the screening tool. I assume this cost decreases in the subjectivity degree of the screening signal according to $d_{\theta} \equiv (1 - c_{\theta})$. Scoring a candidate's written test differently than the publicly-observable signal poses a much higher threat of detection than underscoring someone after an interview because the person did not appear to be friendly or an "appropriate fit". The degree of discretion enabled by a screening tool plays a central role in the model, as it loads on the bias term and determines its relative weight in the evaluator's utility.

²⁸Alternatively, one can model signal precision to depend on group membership (for example, Aigner and Cain (1977), Lundberg and Startz (1983), Cornell and Welch (1996), Bartik and Nelson (2022)). Similar to Autor and Scarborough (2008), I assume it to be independent of group membership to focus the analysis on the new features that I introduce in the model.

²⁹The constant aggregate hiring rate *K* is assumed to be below 50%, as is the hiring rate of each demographic group. This assumption is motivated by the fixed number of positions in the job announcements for public servants, as well as the average probability of being hired being around 4% in the data.

5.2.3 Hiring decision

Workers are hired based on a noisy indicator of productivity generated by screening tool θ :

$$\theta^* = y + \nu_{\theta}(x) + \varepsilon_{\theta}, \quad \varepsilon_{\theta} \sim N(0, 1/h_{\theta})$$

where the productivity signal from tool θ is allowed to be mean-biased by including the bias term $\nu_{\theta}(x)$. This mean-shifter can be interpreted as the disparate impact of the screening tool, which favors men when $\nu_{\theta}(m) > \nu_{\theta}(f)$.³⁰ After observing θ^* , the evaluator updates her assessment of expected productivity of candidates as

$$\mu(x,\theta^*) = \theta \frac{h_\theta}{h_0 + h_\theta} + \mu_0(x) \frac{h_0}{h_0 + h_\theta} + \nu_\theta(x)$$

The expression above represents a weighted average of perceived productivity of group x and the signal provided by the hiring tool, with weights determined by the relative precision of the signal with respect to productivity dispersion. A direct implication from the updated group mean $\mu(x, \theta)$ is that when a screening method is less informative, the evaluator relies more heavily on the group prior.

The hiring decision that maximizes the evaluator's objective function satisfies the rule Hire = $\mathbb{1}\{\mu(x,\theta^*) + d_{\theta}\pi(x) > k_{\theta}\}$, where k_{θ} is the threshold that yields a hiring rate of *K*. The solution to this problem gives the hiring rate $1 - \Phi(z_{\theta}^*(x))$ with hiring threshold $z_{\theta}^*(x)$ for each group. Due to the linear form of the signal expression above, the hiring threshold for group *x* decreases when the group mean productivity is higher, when tool bias favors the group, or when evaluators are biased toward *x* (regulated by the discretion in the tool being used).

5.2.4 Model predictions

To match the model predictions to our empirical setting, consider the screening tool choices available to employers before and after the impartiality requirement. Before the Constitution, employers could use *i*) a written test, which generates a signal *s*; *ii*) a non-written test with signal η ; or *iii*) a combination of both written and non-written tests.³¹ After the requirement, employers in the federal government are constrained to screen candidates using only a blind written test (*s*(*b*)^{*}) or a combination of non-written and blind written tools.

³⁰I consider that the different screening tools provide signals of productivity determined by one factor, which is to say they measure the same skill. A two-factor model would reformulate productivity as $y = y_1 + y_2$, where, y_1 could represent soft skills and y_2 hard skills (similar to Frankel (2021)).

³¹Within the model, I do not distinguish whether employers use one or multiple tests of the same type. A richer formulation that would incorporate the supply of candidates could take into account the number of exams and therefore the length of the screening process as an application deterrent.

This means that making the screening tools already in use more impartial changes the signals used by evaluators from s^* to $s(b)^*$ (full blinding) or from s^* , η^* to $s(b)^*$, η^* (partial blinding). In the cases where the existing mix of screening tools is changed, productivity signals are modified in three potential ways. From an a non-written η^* to blind test $s(b)^*$; from η^* to $(s(b)^*, \eta^*)$; or from (s^*, η^*) to $s(b)^*$.

The full model solution and predictions to gender hiring gaps for each one of the five changes in screening methods is in Appendix C. To summarize the most important insights, I analyze Figure 5, which contains three panels. Panel (a) shows the model-implied gender hiring gap when screening only uses w (gray) or alternatively only uses w(b) (green). The gender gap is plotted as a function of evaluator bias toward a gender. Unless evaluators are biased toward women, the complete removal of discretion reduces the gender gap since it completely eliminates disparate treatment.

Panel (b) compares gender gaps when screening only contains a subjective step to when a job process adds a blind written stage. This plot varies on the horizontal axis the relative tool bias of nw with respect to w(b). Unless the disparate impact of blind written tests is much larger than that of interviews, the increase in screening precision in w(b) + nw leads to smaller gender hiring gaps than nw. Lastly, in the case where interviews are removed in panel (c), gender hiring gaps narrow only if the reduction in total bias from removing nw offsets the loss in total screening precision.

5.3 Empirical Strategy for Multiple Treatments

To estimate the causal impact of each change in screening methods individually, I implement the following triple-difference model:

$$y_{ijt} = \delta_{1} \text{Female}_{i} + \delta_{2} \text{Federal}_{j} + \delta_{3} \left(\text{Post}_{j(t)} \times \text{Federal}_{j} \right) + \delta_{4} \left(\text{Post}_{j(t)} \times \text{Female}_{i} \right) + \delta_{5} \left(\text{Federal}_{j} \times \text{Female}_{i} \right) + \beta_{d} \left(\text{Federal}_{j} \times \text{Post}_{t(j)} \times \text{Female}_{i} \right) + \delta_{o(j)} + \theta_{t} + u_{ijt}$$
(3)

which compares outcomes (y_{ijt}) for female candidates relative to men (Female_i) participating in job processes for the same occupation $(\delta_{o(j)})$ that had screening practices changed following treatment *d*. Standard errors remain clustered at the job process level, as with the previous specifications. Overall, this is a similar model to (1), except that it is implemented separately for each treatment group.

Guided by the theoretical framework in 5.2, each triple-difference coefficient has different interpretations, depending on the conditioning on *d*. To fix ideas for the exposition of results

below, consider the case $d = w \longrightarrow w(b)$. This treatment was experienced by a variety of jobs like school officers and translators whose hiring in the federal government reduced discretion through only blinding the existing written tests. Because the mix of screening tools in use was kept unchanged, overall tool bias and screening precision are not affected by treatment. Under the identifying assumptions, the only effect measured by $\hat{\beta}_{w \longrightarrow w(b)}$ is the response in the gender gap when all sources of evaluator bias are completely removed.

Assessing the plausibility of the identification strategy. I discuss the identifying assumptions and provide evidence in their support next. Similar to the binary treatment design in (1), multiple treatments require the assumption of parallel trends between a treatment and its equivalent control group (the same occupation in state jobs).

One additional consideration in this context is the role of selection into different treatments by occupations with the same pre-existing screening mix. Because treatment assignment decisions were made by high-level bureaucrats at a more aggregated level (occupation) than my unit of analysis (job process), employers and evaluators have limited ability to deviate from top-down guidelines. As a consequence, over 95% of occupations followed one treatment type in job processes across different employers. This results in each treatment being generated by many occupations, but each occupation being assigned to one treatment. Crucially, a wide variety of jobs along several dimensions make up each treatment. These institutional features therefore provide plausible exogenous variation to changes in screening methods.

To investigate this more formally, it is useful to view the institutional portion of treatment assignment as generating a set of potential instruments, where occupations determine changes in screening tools. This is an advantage of my setting since the presence of many occupations in each treatment can be used to leverage overidentification-style tests as in Angrist et al. (2017).

As a first step, consider a set of pre-treatment covariates including occupational skill level and degree of feminization, as well as share of female applicants and competitiveness of the job process. These factors can either directly determine selection decisions into treatment or correlate with variables I do not observe that determine assignment on their own. Table A.8 shows almost no systematic relationship between which treatment is assigned to occupations with the same initial mix of screening tools and these observables.

An additional test addresses the possibility that additional relevant factors in the information set of policymakers may not correlate with the covariates. But to the extent that these unobservables would also matter for gender hiring gaps, they would plausibly lead to pretreatment differences in the outcome paths of jobs that sorted into alternative treatments. For instance, civil engineers and teachers were hired using w + nw before the impartiality requirement and switched to w(b) and w(b) + nw, respectively. If attitudes toward women were different or changing between the two occupations so that the choice of each treatment was based on treatment effects, one would expect pre-treatment outcomes to differ between the two careers.

Pre-trends in outcomes in Figure 6 therefore provide a check for systematic differences across occupations with the same initial set of screening methods. The figure shows dynamic effects plotted from an event-study style version of model (3). Each panel divides occupations according to *pre*, so that they compare treated effects in occupations within subsets (a) w, (b) nw, and (c) w + nw. Across all five treatments, gender hiring gaps between state and federal job processes were not statistically different from zero prior to the new Constitution.

This provides information above and beyond the standard pre-trend inspection test. Gender hiring gaps both followed parallel trajectories, as well as had indistinguishably estimated magnitudes between the pairs of occupations, relative to control job processes. This indicates that, on average, selection of an occupation into a particular treatment shows no relationship with pre-existing differential gender hiring gaps, which in turn would be likely to pick up unobservable pre-treatment factors.³²

Combining the evidence presented and how screening decisions are made in Brazil's public sector, I can test directly for violations of selection into treatment as in Angrist et al. (2017). The Sargan (1958) test, where the endogenous regressor (equivalent to the policy variable $\text{Post}_{j(t)} \times \text{Federal}_j$ in a difference-in-differences specification of (3), ran separately by gender and conditioning for the set of pre-treatment tools, *pre*) is instrumented with occupational codes, imposes an intuitive restriction. If all occupations in *d* adopted that treatment as-goodas randomly with respect to gender disparities in outcomes, one should not expect statistically significant discrepancies of estimated treatment effects between them. Table A.9 confirms this by overwhelmingly failing to reject the overidentifying restrictions for gender gaps in scores and hiring odds as outcomes.

5.4 Main Results

I now proceed to estimate model (3) in five separate regressions for gender final score gaps and gender hiring gaps. Table A.10 shows treatment effects of final scores. For conciseness, I center my discussion in Figure 7, which conducts the same analysis using the gender hiring gap as outcome. Since job offers are solely based on final scores and job openings, any improvement

³²Differences in pre-treatment outcomes between units can be used formally to assess a plausible channel of selection. In Ghanem et al. (2022), who study a binary treatment difference-in-differences setting, large pre-existing differences in the outcome between treated and control make the assumption of parallel trends less likely to hold when there is selection e.g. based on gains. In my setting, the comparison being made is between units receiving different treatments instead of between treated and control units.

in women's hiring rates relative to men's implies a decrease in the average gender final score gap.

Figure 7 analyzes in three groups the five treatment types generated by the impartiality requirement. Each grouping has the same baseline (pre-treatment) screening tool mix *pre* — *w*, *nw*, or w + nw — for which I then estimate treatment effects. Note that, at least observationally, the hiring gap is much larger in job processes relying solely on non-written stages.

Reducing discretion of tools already in use. I first analyze the two treatments that eliminated discretion of tools already in use, without modifying screening stages. With the set of screening tools fixed, overall tool bias (disparate impact) and precision remain unchanged. The two treatments under this category blinded the hiring process either fully ($w \rightarrow w(b)$) or partially, depending on the relative weights between w and nw ($w + nw \rightarrow w(b) + nw$).

Starting with occupations that fully blinded job processes in the federal government, the estimated impact on the gender hiring gap is a decrease of 0.5 percentage point, or about 33% of the baseline gap. This treatment captures the causal effect of removing all sources of evaluator bias, and therefore reveals that they were an important force driving gender disparities.

Now consider the treatment that starts with the screening mix w + nw, blinding the written stage. Due to the presence of non-written steps, the estimated effect may not be similar to the one in the previous treatment. If the written stage has a small weight toward the final score or if evaluators change their scoring in non-written exams as a spillover from blinding, the gender gap may not necessarily narrow. Figure 7 confirms this possibility, indicating that the average treatment effect of partially blinding in hiring processes is null.

With the pre-determined weights at hand, I can test for the two hypotheses in Table A.11, which breaks down the aggregate null estimate by analyzing how gender hiring gaps responded depending on the weight of blind written tests toward the final score. The firs conjecture — treatment effects on hiring gaps should be stronger for hiring processes with greater weight in w(b) — is confirmed. Columns (2), (4), and (8) show that when enough weight is exogenously placed on the written test (at least 50% of the final score), blinding has a positive and significant effect on women's final evaluation scores and hiring probability. Second, the incentive for evaluators to "compensate" men's scores in interviews would be larger the greater the weight placed on written tests, where they can no longer identify candidates. Results in columns (6) and (7) point to no evidence of strategic response by evaluators in non-written tests, which aligns with the previous findings in Section 4 of broadly unchanged performance in non-written stages of women relative to men following treatment.

Adding an objective tool. While a subset of occupations decreased discretion without modi-
fying screening methods, others changed the tools being used. The first treatment of this kind I analyze shifted screening following $nw \rightarrow w(b) + nw$, since relying exclusively on a highly subjective stage would not comply with the impartiality requirement.

This treatment provides lessons to a common approach in other contexts where a standardized, objective tool is introduced to reduce discretion in screening. My theoretical predictions show that adding w(b) increases total screening precision without introducing evaluator bias, since the blind tool gives no discretion to evaluators. Higher screening precision in turn favors minorities since it increases the weight placed on performance signals, as opposed to the (men-favoring) priors. As a consequence, another productivity signal makes the optimal hiring threshold less dependent on the unconditional group mean, reducing the level of statistical discrimination.

On the other hand, a common objection against requiring standardized or written tests is that they could disadvantage women or minorities through a disparate impact channel, if for example women are worse test-takers in a way that is not predictive of worker productivity. But based on my theoretical predictions, as long as they are less biased as a tool relative to interviews, overall bias in screening will still be lower, increasing women's hiring rates. This reinforcing mechanism between bias reduction and better precision leads to large treatment effects in practice. The estimated decrease in the gender hiring gap is about 5.9 percentage points, or 35% of the initial hiring gap from using nw.³³

Removing a subjective tool. The next treatment I analyze also has a pre-treatment mix using both tools, w + nw, but instead of only blinding the written exam, also removes the subjective stage. A mix of high and low-skill occupations — cleaners, programmers, civil engineers, and nutritionists — make up this treatment. Similarly, occupations with the same initial set of tools that kept nw include judges, police officers, and cooks again consistent with discussions on how to modify screening being related to idiosyncratic factors of that job. The motivation for this approach in a broader sense is intuitive: if evaluator bias matters and bias is particularly enabled by tools with high discretion, their use may be an important factor for gender disparities.

My estimates in Figure 7 reveal that removing subjective stages has no effect on the gender gap. This is consistent with a bias-information trade-off where the loss in screening precision from removing a productivity signal offsets the potential gains from eliminating disparate treatment and impact with the use of the tool. Less overall precision in screening

³³Note that in the context of public sector hiring in Brazil, an additional force contributing to a smaller gender gap is a mechanical reduction in the pre-determined weight given to nw, which decreases from 100% to as little as 20% following treatment.

leads evaluators put more weight on their prior, making identifying above-average women more difficult. This result underscores the importance of employers weighing the screening precision loss and net gains from bias reduction when removing a tool.

Replacing a subjective tool. Lastly, I study the treatment where occupations changed the screening design completely — replacing nw with w(b). In this case, occupations traded screening precision and tool bias from non-written exams for those of written tests, with ambiguous predicted effects on the hiring gap. Because of blinding, there is also the complete removal of all sources of evaluators bias. The gender hiring gap decreases by almost 7 percentage points, starting from a baseline gap of almost 17 percentage points. When benchmarked against the initial gap level, the estimated magnitude implies a decrease in the gap of 41%.

Given that removing evaluator bias in nw from the treatment $w + nw \longrightarrow w(b)$ is insufficient to increase female hiring rates, the results for the replacement of subjective tests suggest that either screening precision in written tests is higher than in interviews — since an increase in overall precision is minority-favoring — or that w imposes a smaller disparate impact on women than nw.

Gender effects on sorting. To conclude the main set of results with multiple treatments, Table 4 shows how different changes in screening affect the supply side: job seekers' application decisions. Consistent with the idea from Section 4.3 that perceived discrimination or unfair treatment during hiring may discourage minorities from applying in the first place, columns (1), (4), and (5) all show that blinding a pre-existing written test increases the participation of women in the applicant pool (from 2% to 4% on average).

Column (2) shows that the switch from a non-written exam to a written stage does not affect women's application rates differently than men's. With a completely different screening method, women may perceive the job process as being fairer, but may be uncertain about potentially allocating more time to prepare for the test. In line with a cost of application explanation, column (3) shows that there is also no differential response by gender when employers introduced an additional screening requirement. These results further align with the fact that the reform did not have diversity goals or was promoted as "favoring women", so that applicant responses depends on the specific change to screening methods, with women responding to changes in screening that increase perceived fairness, as long as the hiring process is not seen as more costly. In summary, these results suggests that the way employers screen has important sorting effects and diversity implications.

5.5 Additional Results

In Tables A.12 and A.13, I verify whether estimated effects from different treatments may instead be picking up specific characteristics across occupations. Even though I showed before that each treatment contains a variety job titles and that collectively a number of variables explain a negligible portion of treatment assignment, it could still be possible that a disproportionate presence of say, high skill jobs in one treatment could impact my previous results. The additional set of estimates in both tables, where various occupation characteristics are included as controls show that estimated effects remain unchanged.

While the previous tables show that nothing specific about occupations accounts for the estimated treatment effects, next I investigate potential heterogeneity in treatment effects within each treatment. This is also useful to validate the overidentification test in Section 5.3, as heterogenous effects can confound inference about selection (Angrist et al. (2017)). Table A.14 splits each treatment type into low and high-skill positions and again estimates treatment effects on the gender hiring gap. Overall, results are similar independent of skill level, except in the treatment $w + nw \longrightarrow w(b)$, where for low-skill jobs the estimated effect on the gap is now marginally significant and positive. This is the change in screening where on average the bias-information trade-off was dominated by the loss of screening precision from removing *nw*. The fact that this result flips for low-skill occupations is consistent with screening precision of interviews being lower for low-skill jobs.

6 Conclusion

Hiring decisions shape firm outcomes and determine individuals' access to labor market opportunities. To ensure fair recruiting processes and increase employee gender diversity, organizations face several challenging questions. For example, does replacing interviews with objective tools lead to disparate impact for female applicants? Should screening practices with high discretion be removed, even if they could provide important information about productivity? Causally linking the design of hiring practices to gender equity requires overcoming the lack of data on the decision making process and generating appropriate variation in the choice and implementation of screening methods.

This paper overcomes these challenges and opens the black-box of hiring decisions by developing a natural language processing algorithm that distills high-dimensional, unstructured text records into a uniquely-detailed dataset on the universe of hiring processes in Brazil's public sector. The data contain complete information on candidate performance, including job offers and individual scores, screening methods, and committee hiring members' evaluations of each job applicant in all hiring stages. Combined with the introduction of a requirement that led to a reduction in discretion through different changes in screening methods, my analysis provides important lessons for both private sector firms and professionalized bureaucracies.

Overall, employers can address gender gaps by targeting screening tools used in hiring, without the need to directly change decision makers' fundamental preferences or behavior. However, not all approaches to reduce discretion have the same implications. Screening changes that limit discretion in existing hiring practices or add new impartial screening tools reduce the gender hiring gap by about one third. They also either increase female application rates relative to men, or leave them unchanged when the overall length and cost of participation in a selection process are expanded. But policies that eliminate subjective tools result in a screening precision loss that can outweigh gains from bias removal, having no effects on gender gaps.

Remaining questions to be addressed in future research include the implications of different screening practices for the quality of female and male job seekers. With on-the-job productivity measures that are relevant for the provision of public services, additional work will be able to investigate the potential existence of a equity-efficiency trade-off and its consequences for productivity in the public sector.

References

- Abraham, Lisa, Alison Stein, and J Hallermaier (2020) "Words matter: Experimental evidence from job applications," *Working Paper*.
- Agan, Amanda and Sonja Starr (2018) "Ban the box, criminal records, and racial discrimination: A field experiment," *Quarterly Journal of Economics*, Vol. 133, No. 1, pp. 191–235.
- Aigner, Dennis J and Glen G Cain (1977) "Statistical theories of discrimination in labor markets," *ILR Review*, Vol. 30, No. 2, pp. 175–187.
- Alsan, Marcella, Owen Garrick, and Grant Graziani (2019) "Does diversity matter for health? Experimental evidence from Oakland," *American Economic Review*, Vol. 109, No. 12, pp. 4071–4111.
- Aneja, Abhay and Guo Xu (2022) "Strengthening state capacity: Postal reform and innovation during the Gilded Age," Technical report, National Bureau of Economic Research.
- Angrist, Joshua D., Peter D. Hull, Parag A. Pathak, and Christopher R. Walters (2017) "Leveraging Lotteries for School Value-Added: Testing and Estimation," *The Quarterly Journal of Economics*, Vol. 132, No. 2, pp. 871–919.
- Ashraf, Nava, Oriana Bandiera, Edward Davenport, and Scott S Lee (2020) "Losing prosociality in the quest for talent? Sorting, selection, and productivity in the delivery of public services," *American Economic Review*, Vol. 110, No. 5, pp. 1355–94.
- Atalay, Enghin, Phai Phongthiengtham, Sebastian Sotelo, and Daniel Tannenbaum (2020) "The Evolution of Work in the United States," *American Economic Journal: Applied Economics*, Vol. 12, No. 2, pp. 1–34.
- Autor, David H and David Scarborough (2008) "Does job testing harm minority workers? Evidence from retail establishments," *Quarterly Journal of Economics*, Vol. 123, No. 1, pp. 219–277.
- Baker, Scott R, Nicholas Bloom, and Steven J Davis (2016) "Measuring economic policy uncertainty," *Quarterly Journal of Economics*, Vol. 131, No. 4, pp. 1593–1636.
- Baldiga, Katherine (2014) "Gender differences in willingness to guess," *Management Science*, Vol. 60, No. 2, pp. 434–448.
- Bartik, Alexander and Scott Nelson (2022) "Deleting a signal: Evidence from pre-employment credit checks," *Review of Economics and Statistics, Forthcoming*.

- Behaghel, Luc, Bruno Crépon, and Thomas Le Barbanchon (2015) "Unintended effects of anonymous resumes," *American Economic Journal: Applied Economics*, Vol. 7, No. 3, pp. 1–27.
- Bertrand, Marianne and Sendhil Mullainathan (2004) "Are Emily and Greg more employable than Lakisha and Jamal? A field experiment on labor market discrimination," *American Economic Review*, Vol. 94, No. 4, pp. 991–1013.
- Besley, Timothy, Robin Burgess, Adnan Khan, and Guo Xu (2022) "Bureaucracy and development," *Annual Review of Economics*, Vol. 14, pp. 397–424.
- Bosquet, Clément, Pierre-Philippe Combes, and Cecilia García-Peñalosa (2019) "Gender and promotions: evidence from academic economists in France," *Scandinavian Journal of Economics*, Vol. 121, No. 3, pp. 1020–1053.
- Brands, Raina A and Isabel Fernandez-Mateo (2017) "Leaning out: How negative recruitment experiences shape womens decisions to compete for executive roles," *Administrative Science Quarterly*, Vol. 62, No. 3, pp. 405–442.
- Bybee, Leland, Bryan T Kelly, Asaf Manela, and Dacheng Xiu (2020) "The structure of economic news," *NBER Working Paper*.
- Card, David, Fabrizio Colella, and Rafael Lalive (2021) "Gender preferences in job vacancies and workplace gender diversity,"Technical report, National Bureau of Economic Research.
- Coffman, Katherine B, Manuela Collis, and Leena Kulkarni (2023) "Whether to Apply?" Management Science, Forthcoming.
- Cornell, Bradford and Ivo Welch (1996) "Culture, information, and screening discrimination," *Journal of Political Economy*, Vol. 104, No. 3, pp. 542–571.
- Dal Bó, Ernesto, Frederico Finan, and Martín A Rossi (2013) "Strengthening state capabilities: The role of financial incentives in the call to public service," *Quarterly Journal of Economics*, Vol. 128, No. 3, pp. 1169–1218.
- De Janvry, Alain, Guojun He, Elisabeth Sadoulet, Shaoda Wang, and Qiong Zhang (2023) "Subjective performance evaluation, influence activities, and bureaucratic work behavior: Evidence from China," *American Economic Review*, Vol. 113, No. 3, pp. 766–799.
- Delfino, Alexia (2021) "Breaking gender barriers: Experimental evidence on men in pink-collar jobs."

- Deserranno, Erika (2019) "Financial incentives as signals: experimental evidence from the recruitment of village promoters in Uganda," *American Economic Journal: Applied Economics*, Vol. 11, No. 1, pp. 277–317.
- Doleac, Jennifer L and Benjamin Hansen (2020) "The unintended consequences of ban the box: Statistical discrimination and employment outcomes when criminal histories are hidden," *Journal of Labor Economics*, Vol. 38, No. 2, pp. 321–374.
- Durlauf, Steven N. and James J. Heckman (2020) "An Empirical Analysis of Racial Differences in Police Use of Force: A Comment," *Journal of Political Economy*, Vol. 128, No. 10, pp. 3998– 4002.
- Estrada, Ricardo (2019) "Rules versus discretion in public service: Teacher hiring in Mexico," *Journal of Labor Economics*, Vol. 37, No. 2, pp. 545–579.
- Finan, Frederico, Benjamin A Olken, and Rohini Pande (2017) "The personnel economics of the developing state," *Handbook of Economic Field Experiments*, Vol. 2, pp. 467–514.
- Flory, Jeffrey A, Andreas Leibbrandt, and John A List (2015) "Do competitive workplaces deter female workers? A large-scale natural field experiment on job entry decisions," *The Review* of Economic Studies, Vol. 82, No. 1, pp. 122–155.
- Flory, Jeffrey A, Andreas Leibbrandt, Christina Rott, and Olga Stoddard (2021) "Increasing workplace diversity: Evidence from a recruiting experiment at a Fortune 500 company," *Journal of Human Resources*, Vol. 56, No. 1, pp. 73–92.
- Frankel, Alex (2021) "Selecting Applicants," Econometrica, Vol. 89, No. 2, pp. 615–645.
- Frederiksen, Anders, Lisa B Kahn, and Fabian Lange (2020) "Supervisors and performance management systems," *Journal of Political Economy*, Vol. 128, No. 6, pp. 2123–2187.
- Gentzkow, Matthew, Bryan Kelly, and Matt Taddy (2019) "Text as Data," *Journal of Economic Literature*, Vol. 57, No. 3, pp. 535–74.
- Gentzkow, Matthew and Jesse M Shapiro (2010) "What drives media slant? Evidence from US daily newspapers," *Econometrica*, Vol. 78, No. 1, pp. 35–71.
- Ghanem, Dalia, Pedro HC Sant'Anna, and Kaspar Wüthrich (2022) "Selection and parallel trends," *arXiv preprint arXiv:2203.09001*.
- Gneezy, Uri, Muriel Niederle, and Aldo Rustichini (2003) "Performance in competitive environments: Gender differences," *Quarterly Journal of Economics*, Vol. 118, No. 3, pp. 1049–1074.

- Goldin, Claudia and Cecilia Rouse (2000) "Orchestrating impartiality: The impact of "blind" auditions on female musicians," *American Economic Review*, Vol. 90, No. 4, pp. 715–741.
- Grindle, Merilee S (2012) Jobs for the Boys: Harvard University Press.
- Hoffman, Mitchell, Lisa B Kahn, and Danielle Li (2018) "Discretion in hiring," *Quarterly Journal* of Economics, Vol. 133, No. 2, pp. 765–800.
- Holzer, Harry J, Steven Raphael, and Michael A Stoll (2006) "Perceived criminality, criminal background checks, and the racial hiring practices of employers," *Journal of Law and Economics*, Vol. 49, No. 2, pp. 451–480.
- Hospido, Laura, Luc Laeven, and Ana Lamo (2019) "The gender promotion gap: evidence from central banking," *Review of Economics and Statistics*, pp. 1–45.
- Kingsley, J Donald (1944) "Representative bureaucracy: An interpretation of the British civil service," (*No Title*).
- Kline, Patrick, Evan K Rose, and Christopher R Walters (2022) "Systemic discrimination among large US employers," *Quarterly Journal of Economics*, Vol. 137, No. 4, pp. 1963–2036.
- Kuhn, Peter and Kailing Shen (2023) "What Happens When Employers Can No Longer Discriminate in Job Ads?" *American Economic Review*, Vol. 113, No. 4, pp. 1013–1048.
- Kuhn, Peter, Kailing Shen, and Shuo Zhang (2020) "Gender-targeted job ads in the recruitment process: Facts from a Chinese job board," *Journal of Development Economics*, Vol. 147, p. 102531.
- Lundberg, Shelly J and Richard Startz (1983) "Private discrimination and social intervention in competitive labor market," *American Economic Review*, Vol. 73, No. 3, pp. 340–347.
- MacLeod, W Bentley (2003) "Optimal contracting with subjective evaluation," *American Economic Review*, Vol. 93, No. 1, pp. 216–240.
- McCrary, Justin (2007) "The effect of court-ordered hiring quotas on the composition and quality of police," *American Economic Review*, Vol. 97, No. 1, pp. 318–353.
- Miller, Amalia R and Carmit Segal (2019) "Do female officers improve law enforcement quality? Effects on crime reporting and domestic violence," *Review of Economic Studies*, Vol. 86, No. 5, pp. 2220–2247.

Moreira, Diana and Santiago Pérez (2021a) "Civil Service Reform and Organizational Practices: Evidence from the Pendleton Act," *NBER Working Paper*.

(2021b) "Who Benefits from Meritocracy?" Working Paper.

- Neumark, David (2021) "Age discrimination in hiring: Evidence from age-blind vs. non-ageblind hiring procedures," *Journal of Human Resources*, pp. 0420–10831R1.
- Niederle, Muriel and Lise Vesterlund (2007) "Do women shy away from competition? Do men compete too much?" *Quarterly Journal of Economics*, Vol. 122, No. 3, pp. 1067–1101.
- Ornaghi, Arianna (2019) "Civil service reforms: Evidence from US police departments," Working Paper.
- Oyer, P and S Schaefer (2011) "Personnel Economics: Hiring and Incentives. Volume 4, Part B, Chapter 20 of Handbook of Labor Economics."
- Phelps, Edmund S (1972) "The statistical theory of racism and sexism," *American Economic Review*, Vol. 62, No. 4, pp. 659–661.
- Prendergast, Canice and Robert Topel (1993) "Discretion and bias in performance evaluation," *European Economic Review*, Vol. 37, No. 2-3, pp. 355–365.
- Prendergast, Canice and Robert H Topel (1996) "Favoritism in organizations," *Journal of Political Economy*, Vol. 104, No. 5, pp. 958–978.
- Sargan, J. D. (1958) "The Estimation of Economic Relationships using Instrumental Variables," *Econometrica*, Vol. 26, No. 3, pp. 393–415.
- Shen, Zejiang, Ruochen Zhang, Melissa Dell, Benjamin Charles Germain Lee, Jacob Carlson, and Weining Li (2021) "LayoutParser: A Unified Toolkit for Deep Learning Based Document Image Analysis," arXiv preprint arXiv:2103.15348.
- Small, Mario L and Devah Pager (2020) "Sociological perspectives on racial discrimination," *Journal of Economic Perspectives*, Vol. 34, No. 2, pp. 49–67.
- Xu, Guo (2021) "Bureaucratic Representation and State Responsiveness during Times of Crisis: The 1918 Pandemic in India," *Review of Economics and Statistics*, pp. 1–29.

Figures and Tables



Notes: This shows the stylized structure of a hiring process in the Brazilian public sector posted in raw government publications (official gazettes). Information at the top (dark green) describes the screening dynamics from the moment a job is announced until job offers are sent out. The lower part of the figure (light green) shows variables I construct based on observable information in the text of official government documents. The procedure for data extraction is described in Section 3.

FIGURE 1: Stylized Structure of Public Sector Hiring Processes



(a) Announcement

(b) Applicants

(c) Final Results

Notes: This figure shows screen captures of the raw government gazette pages containing the **(a)** announcement, **(b)** list of applicants, and **(c)** final results for a job process in the federal government to select public prosecutors. Pictures are trimmed and additional posts containing exam scores before the final results are not shown for visualization purposes. Text snippets bounded in red are extraneous posts unrelated to the target hiring process, while text bounded in green represents the correct contents. Shaded areas represent target semantic relationships within text snippets, corresponding to the attributes sender, receiver, feature, and numeric elements.

FIGURE 2: Illustration of Linkage and Data Extraction of Job Processes



Notes: This figure shows the average share of job processes with subjective rounds, objective rounds, and the number of screening rounds used by federal (treated) and state (control) governments. Black boxed boundaries represent pre-treatment averages, while shaded areas plot the same statistic after treatment. All statistics control for occupation fixed effects.

FIGURE 3: Compliance with the Impartiality Requirement





$$\mathbb{P}(\text{Female} = 1) = \delta_{i(o)} + \theta_t + u_{ijt}$$

separately for control (state governments) and treated (federal government) groups. *i* denotes a job applicant, *j* denotes a job process, and *o* denotes the occupation or job title. 1986 is the omitted year. The figure on the right plots $\widehat{\gamma}_0$ estimates of the regression

Final Score_{*ijt*} = $\delta_{j(o)} + \gamma_0$ Female_{*i*} + u_{ijt}

for control (state governments) and treated (federal government) groups separately and for each year. Standard errors are clustered at the job process level across all specifications. Shaded areas are 95% confidence intervals.

FIGURE 4: Dynamic Estimates of Gender Application and Score Gaps



Notes: The figure illustrates predictions for the gender hiring gap using the closed-form solutions in Appendix C. Panel (a) shows the model-implied gender hiring gap when screening only uses written w (gray) or alternatively only uses nonwritten w(b) (green) tools. The gender gap is plotted as a function of evaluator bias toward a gender. Unless evaluators are biased toward women, the complete removal of discretion reduces the gender gap since it completely eliminates disparate treatment. Panel (b) compares gender gaps when screening only contains a subjective step to when a job process adds a blind written stage. This plot varies on the horizontal axis the relative tool bias of nw with respect to w(b). Unless the disparate impact of blind written tests is much larger than that of interviews, the increase in screening precision in w(b) + nw leads to smaller gender hiring gaps than nw. Panel (c) shows the case when nw is removed, showing that gender hiring gaps narrow only if the reduction in total bias from removing nw offsets the loss in total screening precision.

FIGURE 5: Predicted Effects on the Gender Hiring Gap from Different Changes in Screening



Pre-Treatment Mix: w + nw

Notes: This figure compares gender hiring gaps in occupations that had the same pre-treatment screening methods and that followed different treatments. The first panel shows job processes transitioning from $w \longrightarrow w(b)$. The second panel shows gender gaps for occupations that followed $nw \longrightarrow w(b)$ or $nw \longrightarrow w(b) + nw$. The third panel compares outcome paths for the case $w + nw \longrightarrow w(b) + nw$ to $w + nw \longleftrightarrow w(b)$. Shaded areas are calculated with standard errors clustered at the job process level.

FIGURE 6: Gender Hiring Gap Paths in Each Treatment Type



Notes: This figure plots treatment effects estimated from the triple-difference model:

$$y_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) \\ + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) \\ + \beta_d \left(\text{Federal}_j \times \text{Post}_{t(j)} \times \text{Female}_i \right) + \delta_{o(j)} + \theta_t + u_{ijt}$$

which compares outcomes (y_{ijt}) for female candidates relative to men (Female_{*i*}) participating in job processes for the same occupation $(\delta_{o(j)})$ that had screening practices changed following treatment *d*. Treatments are displayed at the top with their corresponding estimates in the bars displayed below. Bars are 95% confidence intervals obtained with standard errors clustered at the job process level.

FIGURE 7: Treatment Effects of Changes in Screening Tools: Gender Hiring Gap

	Con	trol	Trea	ted
	Before Reform	After Reform	Before Reform	After Reform
Education Requirement				
<high school<="" td=""><td>47%</td><td>22%</td><td>13%</td><td>7%</td></high>	47%	22%	13%	7%
High School	26%	17%	27%	8%
College or more	27%	61%	60%	85%
Screening Steps				
Average # Rounds	1.9	1.7	1.3	2.1
% Rounds Objective	47.4%	43.3%	64.4%	79.2%
% Rounds Subjective	39.4%	43.8%	21.9%	13.6%
% Rounds Resume	13.2%	12.9%	13.7%	7.2%
Joh Annlicants				
Avg # Applicants	34 7	33.6	34.4	183
Total # Applicants	34 871	<i>4</i> 1 736	18 7 2 6	12 600
	0.400	41,730	10,720	12,000
# Job Processes	2,422	838	1,005	2,289

 TABLE 1: Selected Characteristics of Job Processes

Notes: These are descriptive statistics of job processes published during 1986-1990. Treated job processes are those in Brazil's federal sector. The control group comprises processes in the states of Amazonas in the country's north region, Pernambuco in the northeast, Distrito Federal, Mato Grosso, and Mato Grosso do Sul in the central region, São Paulo — the largest and richest state — in the southeast, and Rio Grande do Sul in the south. Sample statistics for *Screening Steps* are net of occupation fixed effects to compare temporal changes within different hiring processes for the same job title.

	Final Score			W	Written Score			Non-Written Score		
	Women (1)	Men (2)	(3)	Women (4)	Men (5)	(6)	Women (7)	Men (8)	(9)	
$\operatorname{Federal}_{j} \times \operatorname{Post}_{j(t)}$	0.067** (0.030)	-0.075* (0.037)		0.024 (0.044)	-0.109* (0.059)		-0.010 (0.071)	0.020 (0.099)		
$\operatorname{Federal}_{i} \times \operatorname{Post}_{i(t)}$			0.141***			0.134*			-0.031	
\times Female _i			(0.048)			(0.074)			(0.122)	
Pre-treatment mean	-0.04	0.07	-0.12	-0.02	0.03	-0.05	-0.02	0.13	-0.15	
Occupation FE	Х	Х	Х	Х	х	Х	Х	Х	х	
Year FE	Х	Х	Х	Х	Х	Х	Х	Х	Х	
Obs.	54,892	32,067	86,959	34,511	15,546	50,057	29,444	10,764	40,208	

TABLE 2: Estimates of Screening Impartiality on Candidate Scores

Notes: The table shows difference-in-differences estimates of outcomes for applicant *i* in job process *j* in a given year *t*. The first specification is $y_{ijt} = \delta_2 \text{Federal}_i + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_{j(o)} + \theta_t + u_{ijt}$ only with female candidates in columns (1), (4), and (7), only with male candidates in columns (2), (5), and (8). The second specification for columns (3), (6), and (9) is $y_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(o)} + \theta_t + u_{ijt}$. The outcome *y* represents either a candidate's final score, written score (written exams), or non-written score (interview, oral, practical exams). If a job process occurred $t \ge 1989$, $\text{Post}_{j(t)}$ is a dummy equal to 1, and if it was for a federal job, $\text{Federal}_j = 1$. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

	$\mathbb{P}(Hired \text{Female} = 1)$ (1)	$\mathbb{P}(Hired \text{Male} = 1)$ (2)	$\mathbb{P}(Hired \text{Applied} = 1)$ (3)	$\mathbb{P}(\text{Female} = 1)$ (4)
Federal _j × Post _{j(t)}	0.003**	-0.004***		0.010**
	(0.001)	(0.001)		(0.005)
$\text{Federal}_{i} \times \text{Post}_{i(t)}$			0.007***	
\times Female _i			(0.002)	
Pre-treatment mean	0.09	0.11	-0.015	0.52
Occupation FE	Х	Х	Х	Х
Year FE	Х	Х	Х	Х
Obs.	54,892	32,067	86,959	86,959

TABLE 3: Estimates of Screening Impartiality on Hiring and Application Rates

Notes: The first column shows a regression coefficient capturing the probability that a given female job applicant receives a job offer: $\mathbb{P}(Hired = 1)_{ijt} = \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_{j(o)} + \theta_t + u_{ijt}$ which is ran only on female individuals, column (2) runs the same regression with male applicants, column (3) runs $\mathbb{P}(Hired = 1)_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(o)} + \theta_t + u_{ijt}$ which measures the probability female job applicants receive an offer relative to male job applicants — the gender hiring gap. Finally, column (4) regresses the specification $\mathbb{P}(\text{Female} = 1)_{ijt} = \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_{j(o)} + \theta_t + u_{ijt}$. If a job process occurred $t \ge 1989$, Post_{j(t)} is a dummy equal to 1, and if it was for a federal job, Federal_j = 1. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

		% Female Applicants									
	w o w(b) (1)	nw o w(b) (2)	$nw \to w(b) + nw$ (3)	w + nw o w(b) (4)	$w + nw \to w(b) + nw$ (5)						
$\operatorname{Federal}_{j} \times \operatorname{Post}_{j(t)}$	0.023*** (0.008)	-0.004 (0.013)	-0.022 (0.020)	0.054*** (0.009)	0.043*** (0.012)						
Occupation FE	Х	Х	Х	Х	Х						
Year FE	Х	Х	Х	Х	Х						
Obs.	1,145	900	1,822	4,252	3,106						

TABLE 4: Treatment Effects of Changes in Screening Tools: Share of Female Applicants

Notes: This table plots treatment effects for each treatment type *d* using the regression:

% Female Applicants_{*jt*} =
$$\delta_2$$
Federal_{*j*} + $\delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_{o(j)} + \theta_t + u_{ujt}$

where $\text{Post}_{j(t)}$ is a dummy equal to 1 if a job process occurred $t \ge 1989$ and $\text{Federal}_j = 1$ if it was for a federal job. Treatment type *d* represents job process transitioning from written exam to blind-written exam $(w \longrightarrow w(b))$, a job process comprising a non-written exam switching for a blind-written $(nw \longrightarrow w(b))$, or only adding the blind-written test $(nw \longrightarrow w(b) + nw)$, and a hiring process using a mix of written and non-written tools potentially dropping the non-written and blinding the written $(w + nw \longrightarrow w(b))$ or just blinding the written $(w + nw \longrightarrow w(b))$ or just blinding the written $(w + nw \longrightarrow w(b))$ or just blinding the written and non-written and potentially dropping the non-written and blinding the written ($w + nw \longrightarrow w(b)$) or just blinding the written ($w + nw \longrightarrow w(b)$) or just blinding the written ($w + nw \longrightarrow w(b)$) process and have standard errors are clustered at the job process level.

Designing Gender Equity: Evidence from Hiring Practices

INTERNET APPENDIX FOR ONLINE PUBLICATION

Appendix A Figures and Tables



Notes: This figure shows the history of all changes to the selection process of public servants in Brazil, beginning from when the country was still under Portuguese domain, and spanning democratic and military control periods. Brazilian legal experts and historians consider the 1934 Constitution (amended in 1939) to establish meritocratic public servant selection — one of the first countries in Latin America. This early stage, however, provisioned the use of examinations or titles (resume) for some occupations. The 1946 Constitution expanded the selection criteria for most government jobs, until in 1967 and 1969 under military regime, the selection of every public servant through the legal device known as *Concurso* had to include at least one type of examination, ruling out the sole use of resumes. Despite the language, the definition of examination at that moment was fairly broad, so that interviews would be character or personality "exams", for example. In the end of 1988, Brazil passed a new Constitution which kept all public servant selection criteria from the previous Constitution but required public sector job processes to be conducted impartially. I exploit the introduction of this requirement as the main source of variation for part of the empirical analysis in the paper.

FIGURE A.1: History of Changes in the Selection of Public Servants in Brazil

4.5. As provas escritas e pratica terao a duração de O4 (quatro) horas, cada uma, e, na prova oral, não excederá de 45 (quarenta e cinco) minutos para cada candidato, sendo esse tempo dividido, proporcionalmente, entre os membros da Comissão Examinadora.

4.6. Durante a realização das provas é proibido o uso de quaisquer anotações, facultada a consulta a textos legais, desde que sem comentários ou notas explicativas, exceto quanto à primeira prova, quando nenhuma consulta será permitida.

4.7. Não haverá segunda chamada para qualquer das provas. 4.8. Não será admitido em sala o candidato que comparecer borário estabelecido.

4.9. Será excluido do concurso o candidato que faltar a qualquer das provas, <u>que as tornar identificaveis</u> ou que, durante a realiza ção delas, comunicar-se com outro candidato ou com pessoas estranhas, oralmente ou por escrito, ou, ainda, que se utilizar de notas, impre<u>s</u> sos ou livros, salvo os textos legais permitidos.

anós

4.10. 0 canuldato, ao entregar a prova, receberá comprovante de seu comparecimento.

Notes: Selection Process Rules for Hiring Federal Judges (Sep 4, 1989). Reads as: "*Candidates identifying themselves in any exam will be excluded from the hiring process.*"

FIGURE A.2: Enforcing Blind Exams After Reform

nº 120-	E, sexta feira, 22 de	junho de	2001		di	irið?	biid	äρ.			. s	eção 3	3!	-
007772-12	SIRLEI DE CASTRO ARAU-	p1 [01-09	2001 14:00	011406-06	CELSON R	OBERTO CAN	moja j	16-07-2001	08:00	003481-11	ANA CEL	I RODRIGUES DA	11 30-07-3	2001 08:00
001186-04	JO VALERIA NOGUEIRA DE	31 01-08	-2001 14:00	001005-05	FRANCISCO	JORGE CIV	VI-31	16-07-2001	14:00	007595-04	ANDREA	CARLA SKRABA	11 30-07-0	00:80 100
Região Sul	SOUZA Biologia Celular e Cultura de			003315-01	JOSIANE 1	ERESINA CA	NR-31	16-07-2001	14:00	004922-09	ANTONIO	PASQUALETTO	11 30-07-5	00:80 100
008702-06	ADRIANA TOURINIIO SA-	31 25-07	-2001 08:00	011537-07	JOSUE SAN	TANA PIMTO	31	16-07-2001	14:00	006927-10	ARY GE	RTES CARNEIRC	30-07-3	00:30 100
003561-13	CARLA SIMONE DE ALMEI- DA CUALA	31 25-07	-2001 08:00	003321-07	MARCELO	GALEAZZI C	24-31	16-07-2001	14:00	009186-09	EDUARD	GIOVANININI	U 30-07-3	801 14:00
04150-05	CHRISTIANE CONTIGLI DILMA CRISTINA ANGRA	31 25-07 31 25-07	2001 06:00	003223-13	MARION D) ROCIO FOER	RS-31	17-07-2001	14:00	029971-01	FABIO NO	LASCO	11 30-07-	1001 14:00
04414-08	EDGARD AUGUSTO DE TO- LEDO PICOLI	31 25-07	-2001 08:00	003658-06	PAULO CES REGINA L	AR BOGORNI UCIA SUGAY	31	17-07-2001	08:00	000403-11	NI DR ME	NEZES JUNIÓR	11 30-07-0	1001 14:00
03921-09	JOAO CARLOS BESPALHOK FILIO	31 25-07	-2001 08:00	028372-01	MA BONALDO	REIS JUNIOR	31	17-07-2001	08:00	002501-03	GUSTAVO	SCHIEDECK	11 31-07-3	00:80 100
11095-07	LIANE TEREZINHA DORNE- LES	31 25-07	-2001 08:00	003920-08	ROSQUI BA	IBEIRO CHOO REOZA	20-31	17-07-2001	00.90	010676-04	JOAO BA	TISTA BELTRAC	11 31-07-2	00:80 100
01414-05	CIUNCULA	31 25-07	2001 14300	003326-05	ALMEIDA P	ERES	21	17-07-2001	14:00	003620-07	JOAO CAE	TANO FIORAVAN	11 31-07-3	00:80 100
011178-12	NELSON BARBOSA MA-	31 25-07	2001 14:00	Região Sul	Fitopatologia	M.100052		0.50.50001	19.00	003644-05	JOAO LEC	NARDO FERNAN-	11 31-07-2	001 14:00
10738.4-13	PAULO CELSO DE MELLO	31 25-07	2001 14:00	003535-13	ADRIANE #	ARIA DELG	iA- 11	16-07-2001	03:00	010569-01 003236-13	JOEL DON JOSE ANT	AZZOLO ONIO SCHAMNE	11 31-07-2	001 14:00
003584-10	PEDRO CANISIO BINSFELD	31 25-07	2001 16:00	011074-12	ADRIANE W	UNDLAND	0	16-07-2001	08:00	011050-01	JOSE ER	ANI SCHWENG-	11 31-07-2	001 14:00
Região Sul	Biologia Molecular Animal			005426-06_	GERNARDO	UENO IARCIA DUAR	TEU	16-07-2001	08:00	003611-11 003661-09	LEANDRO	MARIA MULLER WARGAS	11 31-07-2	001 14:00
006620-02	ANTONIO HERNANDES TORRES JUNIOR	23 24-07-	2001 08:00	008597-02	RIOS FARIA CELIA RE	GINA TREM	IA- 11	16-07-2001	14:00	003642-03 008559-06	MARIA III	LÉNA FERMINO	11 01-08-2	001 08:00
008973-04	CARLOS ALEXANDRE SAN- CHEZ FERREIRA	23 24-07-	2001 08:00	003501-02	COLDI CLAUDIA V	TEIRA GODOY	(11	(6-07-2001	14:00	003220-10	MOACIR ROLT	ROBERTO DA-	11 01-08-2	001 08:00
02805-12	LIMÁ ROSA	23 24-07	2001 08:00	009909-04	SEIXAS	DINALI SANTO	osii	16-07-2001	14:00	010329-09 008713-04	PAULO JO	SE DE OLIVEIRA	11 01-08-2	001 14:00
03565-05	DANIELA ALMEIDA FREI-	23 24-07-	2001 08:00	008243-02	MACIEL	DATO NUNI	BS 11	16-07-2001	14:00	011767-03	REGINALE	O TEODORO DE	11 01-08-2	001 14:00
03634-11	TAS ERICA_ELIAS_BARON	23 24-07-	2001 (08:00	003406-01 008338-06	LEANDRO L	ARTINS MORE	BI-11	17-07-2001	08:00	008434-11	RENATO	VASCONCELOS	11 01-08-2	001 14:00
07503-03	LAURA ISABEL WEBER DA	23 24-07-	2001 14:00	003904-05	LUIZA HEL	ENA KLINGE	a. 11 1	7-07-2001	08:00	010585-04	VALTER B	OBERTO SCHAF-	11 01-08-2	001 14:00
10425-13	LUCIANA RELLY BERTOLI-2	23 24-07-	2001 14:00	008348-03	MARCIA SQ	ARES CHAVES	s_u	7-07-2001	03:00	Região Sul	Fecnologia	de Sementes e Mu-		
02813-06	MAGDA VIEIRA BENAVI-2	23 24-07-	2001 14:00	007532-06_	MARCIN	IKQ NISHUM	10 1	7-07-2001	14:00	003497-01	ANGELICA	BROB RODO	11 30-07-2	00:00 100
11813-10	MARIA CRISTINA CABRAL 2	23 24-07-	2001 14:00	002927-03	RENATA GA	KUKA SUARD		7-07-2001	14:00	000040-13	DE ANDE/	DE DE OLIVEIRA	11 30.07.2	001 08.00
03424-06	MOISES BASILIO DA CON-2	23 24-07-	2001 14:00	010333-12	NEZI	SUZZO BASSA	A-111	7-07-2001	14000	011409-09	CECI CAST	TILIO CUSTODIO	11 30-07-2	001 08:00
2825-05	ROBERTA MATTOS COLLA-2	23 24-07-	2001 14:00	007553-01	ROBSON M/	ACELO DI PIL	15-11 I	7-07-2001	14,00	029000-05	DOS	ECIAK DOS SAN-	30-07-2	005 14:00
ela, 86.67	/ 10000434, Guilherme Morei	ira Loures da	Costa, 90.00	2.1.2	Convocação	dos candidatos	s sub judice	e para a av	aliação de	titulos e para a	sindicâne	ia de vida pregres	sa, observado o	disposto
Santos L 01102 Ja	ima, 99.34 / 10001742, Italo (Goncalves M	522, nerberi jura, 89.67 /	no subitem 12. na seguinte or	l do Edital n dem: número	* 1/2013-BCB/ de inscrição r	PGBC-DEI e nome do	eandidato	de agosto i em orderr	1e 2013, publicac alfabética.	lo no Diár	io Oficial da União	o de 7 de agoste	de 2013,
es Ferreis 84 / 10001	ra, 95.67 / 10006319, Joao 4 1460, Joao Lucas Saldanha e M	Gabriel Ville firanda, 87,50	a Machado, / 10003714.	10008 3 DA	210, Fernand AVALIACĂ	la Batista Matt O DE TÍTULA	tos / 10001 OS	083, Mau	reen Cristi	na Sansana / 10	003937, V	ictor Hugo Lopes	Antunes.	
Nunes (le Viveiros Neto, 88.17 / 1000 10006416, Jorge Maffra Ottor	00510, Joao V ni, 93.67 / 10	fictor Moura 103701, Jose	3.1 0	s candidatos	convocados pa	ara a avalia	ção de titu	llos dispor	ão dos dias 17 e	18 de ma	rço de 2014, no h	orário das 8 ho	rras às 12
Lima Cou 17 / 1000	nto Neto, 93.34 / 10000689, J 1002, Kamila dos Santos Tabi	Juliana Buena aquini, 88.83	Bergmann, / 10007425,	Cided	ALLE ALLE	inces (incento	o rocarj, pa	ra a conce	pa une una	Local/andars	enocieços	nistatore a seguit.		
e Aviia P 004332, 1 1055 East	etry, 83.00 / 10002244, Lana . Leandro Figueiredo Silveira, 9	0.34 / 100039	Malo Onin	Belém/PA	001	Escola Meu Pe - Nazaro	edacinho d	o Céu - Ri	a Boaven	tura da Silva, nº	1.004 (ent	re Generalissimo I	Deodoro e 14 d	e Março)
lha Junic 04442 La	ar, 85.33 / 10006836, Letici acas Antunes de Signeira Cost	ia Maia Via ta 77.67 / 10	na, 87.17 / 009722 La-	Belo Horizont	e/MG	Faculdade Piti Jardim	ágoras de l	Belo Hori:	conte - Ur	tidade Cidade Ja	rdim - Ru	a Santa Madalen	a Sofia, nº 30	- Cidade
Soares Ba 50 / 10001	umfeld, 94.17 / 10001156, Lu 1819, Marcela Freitas Costa, 8	iz Felipe Hor 3.67 / 100007	owitz Lopes, 64, Marcella	Curitiba/PR		norte, mezanir Colégio Estad	no - Asa N Jual Paulo I	lorte Leminski -	Rua Core	mel Augusto de	Almeida 4	Garret, nº 135 - T	arumă	CC), ala
tschbach, 51 / 1000	86.34 / 10005888, Maria Bea 2355, Mariane Streisky Bitter	atriz de Olive ncourt, 88.17	ira Fonseca, / 10003889,	Fortaleza/CE	0.0	Instituto Muni Damas	icipal de Pe	rsquisas, A	dministraç	ão e Recursos H	umanos (I	MPARH) - Avenis	la João Pessoa,	nº 5.609
lissa Krel tscher, 92.	ing, 87.84 / 10004777, Milto 17 / 10004465, Murilo Santos I	n Guilherme Ramos, 87.84	de Almeida / 10006500,	Recife/PE	1.5	Farroupilha Faculdade Est	tácio de Sá	i do Recit	e (antiga	FIR-NPJ) - Prés	s da Cun fio Anexo	1.771 - Avenida	Engenheiro A	h 527 -
drade, 80.	rda Macedo Costa, 90.34 / 10 17 / 10003217, Nicole Campos	5 Costa, 87.67	/ 10007723,	Rio de Janeiro	×RJ	Carvalho, nº 1 Associação Cr	1.771 - Ma arioca de E	dalena insino Sup	erior - UN	ICARIOCA - A	venida Pa	alo de Frontin, nº	568 - Rio Con	tprido
ujo, 91.8	4 / 10006704, Pedro Nazare	de Mendon	a Procopio,	São Paulo/SP		Faculdades Mo	letropolitana	as Unidas	(FMU) - I	Prédio 6 - Rua T	aguá, nº 1	50 - Metró São J	ooquim - Liber	dade
fael Camps Castro 9	as de Faria, 89.50 / 10001785, 44.00 / 10005209. Rafael Ge	Rafael Cava	Icanti Garcia											
.01 / 1000 fael Nogu	2591, Rafael Henrique Fortu eira de Lucena, 85.17 / 100	anato, 94.17 01448, Rafac	/ 10002555, l Rodrigues	3.2 Pi todas as instr DCD/DCDC/D	ara a avaliaçã ruções contis	io de titulos, o das no item	12 do Ec	deverà ob fital nº l	servar /2013-	I. Justificativa: Valor R\$3.740,0	0 Ratifica	nto por representa ção: Maria Apare	nte comercial e cida Ferraz de	menezes.
ssoa de Ma Albuquerqu	elo Camara, 96.17 / 10008584, ac, 90.34 / 10001048, Ravi de	Raissa Carva Medeiros Pei	alho Fonseca xoto, 87.33 /	3.3 R	loceberá nota	zero o candi	lidato que	não entrej stitul de al	gar os	Cnete Aujunta (IS Density	DOS DE HABII	1014. ITACÃO	
meida Bor	enata Almeida de Moura, 893 n, 79.83 / 10000074, Robert	to Alves Gor	2, Rhini de nes, 85.67 /	e neste edital. 4 DA	SINDICÂN	CIA DE VIDA	A PREGRE	SSA		CONCORRÉ	NCIA IN	TERNACIONAL	N ¹ 133/2013-E	DEMAP
Examin Didatica - M	ador - Projeto - Argaição do Me ledia Ponderada.	emorial - Prova	Bealiz	w-se nos dias 26 e	27 de maio d	se 2008, nesta	de 04 de romero 4	setembro di 10276, ao di	e 2007, can ual estava in	go nomero 108730 Iscrita a Professora	4 / claro Doutora	ProP Dr ^a Ana N ProP Dr ^a Rosan	Aarcia Spano Naka e Pikt Pessa Ribei	no: 9,5 - 9,5 - ro: 9,5 - 9,5 -
Prof. Dr Prof. Dr	. Bernardo Mantovani: 10,0 - 10,0 - . Joaquim Coutinho Neto: 10,0 - 10,	- 10,0 - 10,0 1,0 - 10,0 - 10,0	Facuidade, 01 (um) can	Concurso de Titulos o de Professor Douto	e Provas para o or, referência Mô	provimento de S-3, em Regime	Maria do I A Cor	Carmo Gulla missão Julga	ci Guimarăe dora atribui	s Caccia-Bava. u as seguintes notas	a candi-	A Congregaçã em 24 de junho de 1	o em sua 742º Se 2008, homologou	ssão Ordinária o Relatório Fin
Prof. Dr Prof" Dr	r. Carlos Curti: 10.0 - 10.0 - 10.0 - 1 " Ana isabel de Assis Pandochi: 10.0	0,0 0 - 10,0 - 10,0 -	de Dedicaç junto ao De	o Integral à Doctri partamento de Fisiol	icia e a Pesqui logia - Disciplina	sa (R.D.I.D.P.), as de Fisiología	data: Exami	inador - Pror	a Pratica - I	Prova Didática - Arg	niçao do	FACULDADE D	E MEDICINA DE R	IBEIRÃO PRET
10,0 ProP Dr	^e Maria de Lourdes Teixeira de Mora	es Polizeli: 10,0	- RCG-212 Neurofisiok	Estrutura o Função gia, referente ao Ei	dio sistema Ner dital FMRP-USP	voso - Area de P nº 046/2007,	Memorial Prof. I	- Média Pon Dr. Juan Stu	derada. Indo Yazile F	locha: 10,0 - 9,8 - 1	0.0 - 9.9	COMUNICADO Realizou-se n	FMRP-USP Nº 05 os dias 09 e 10 c	9/2008 Je junho de 2
- 10.0 - 10.0 A Con	gregação em sua 742º Sessão Orda	inária, realizada	2007, cargo	número 1087037 /	Listado de 1º d claro número 4	e setembro de 15529, ao qual	Prof. I	Dr. José Sebi	stião dos S	antos: 10,0 - 9,5 - 9, antos: 10,0 - 9,8 - 9	8 ,7 - 9,8	01 (um) cargo de Pr	so de Titulos e Pro ofessor Doutor, ref	vas para o pro orência MS-3,
do pela Con	nissão Julgadora. DADE DE MEDICINA DE DIREIRÃO R		Merari de F	tima Ramires Ferrari	, Andressa Daia	ne de Carvalho	ProP	Dr ^a Hillegon	da Marai Du	lith Novaes: 9,9 - 9	8 - 9,7 -	junto ao Departame	nto de Farmacolo	ja - Disciplina
COMU	NICADO FMRP-USP Nº 046/2008 ou-se nos días 28 e 29 de maio o	de 2008, nesta	As car Fatima Ran	didatas Fabiola Cris res Ferrari, Andressa	tina Ribeiro Zu Dajane de Carr	chi, Merari de valho comunui-	A Co em 24 de j	ingregação e junho de 20	im sua 742º 38. homolog	Sessão Ordinária, ou o Relatório Final	realizada elabora-	FMRP-USP Nº 096/2 de 05 de setembro	007, publicado no de 2007, claro/ca	Diario Oficial no namero 10
Faculdade, o 01 (um) card	o Concurso de Titulos e Provas para o go de Professor Doutor, referência M	o provimento de IS-3, em Regime	caram prev presente co	amente as suas der icurso.	sistências em c	oncorrerem ao	do pela Co FACL	unissão Julg JUDADE DE N	adora. MEDICINA D	e Ribeirão Preto		qual estavam inscri Andrade, Márcio Au	tos: Daniel Fernar qusto Fressatto de	des, Claudia I Godoy, Maru
de Dedicaçã junto ao De	ão Integral à Doctincia e à Pesqui partamento de Bioquímica e Imuno	itsa (R.D.I.D.P.), xlogia - Area de	A Con didato:	ssão Julgadora atrib	uiu as seguinte	s notas ao can-	COM Real	UNICADO FI izou-se nos	dias 02 e 0	055/2008 13 de junho de 200	R. nesta	Daniella Bonaventu Matsuda, Joice Ma	ra, Valéria Cristin ria da Cunha, Le	ia Sandrim, N ionardo Resst
Bioquímica Edital FMRF	 Disciplina de Bioquímica - RCG 100 P-USP N* 035/2007, publicado no D 	08, referente ao Jiário Oficial do	Examin Memorial -	idor - Conferência - Média Ponderada.	- Prova Didatia	- Argüição do	Faculdade 01 (um) ca	, o Concurso irgo de Profe	de Titulos e ssor Doutor,	Provas para o provi referência MS-3, er	nento de n Regime	Moraes, Marcelo M Andréa Sanchez, C	artins Reis, Waldio Tarolina Barakli A	au Aparecido V raŭjo Restini
Estado de O claro núme	16 de setembro de 2007, cargo non ro 1085514, ao qual estava inscri	nero 1085518 / ito o Professor	Prof ^o D Prof. D	Anette Hoffmann: Wamberto Antonio	10,0 - 10,0 - 10 Varanda: 10,0	.0 - 10.0 - 10.0 - 10.0 -	de Dedica junto ao D	ição Integra Separtament	l a Docenci o de Neurol	a e a Pesquisa (R. ogia, Psiquiatria e P	DJ.D.P.). sicologia	Fiacadori Tavares. Nao comparer	eram ao refeirdo	concurso os o
A Com	celo Damário Gomes. issão Julgadora atribuiu as seguinte	is notas ao can-	10,0 Prof. D	Fernando Morgan o	de Aguiar Corre	e: 10.0 - 10.0 -	Médica - Condições	Disciplina Fisicas, di	de Terapia e acordo co	Ocupacional Apl om o Edital FMRS	cada as NUSP nº	Daniel Fernandes, Claudia Roberta di	Marucia Chacur, a Andrada, Marci	Marcelo Ma o Augusto Fr
Examin	ador - Projeto - Argüição do Me	morial - Prova	10.0 - 10.0 Prof. D	Helio Zangrossi Jun	lior: 10,0 - 10,0	- 10,0 - 10,0	setembro	de 2007, c	no Diario (argo númer	o 1087355 / claro	número	Godoy, Joice Man Flacadori Tavares.	a da Cunha, An	rea sanchez
Prof. Dr	iedia Ponderada. I. Bernardo Mantovani: 10,0 - 10,0 -	10.0 - 10.0	A Con	regação em sua 74	2ª Sessão Ordir	naria, realizada	A15510, a Meirelles (io quai estar Carril Elui,	la inscrita a	Professora Doutor	a vakna	Candidata Dani	igadora atribulu a iella Bonaventura:	s seguintes no
Prof. Dr	r. Carlos Curti: 10,0 - 10,0 - 10,0 - 10, Manuer Familia des Sector: 10,0 - 10,	0,0	do pela Cor	istão Julgadora.		OCTO	data:	inssao Julga	o Dettice	as seguntes note	nia canu-	Didatica - Media Po	nderada.	10.0 0.5 0.0
10,0 Profit Dr	r. wagner Ferreira dos Santos: 10,0 1 Maria da Lourdor Toixoira da Mora	or Bolizelii 10.0	COMU	NICADO FMRP-USP 1	Nº 051/2008	KEIU	Memorial Drof	- Media Pon	derada. Notucor Dar	nova Diceoca - Avç	2 9 4	Prof [®] Dr [®] Maria	Cristina de Oliveir	a Salgado: 9,8
10,0 - 10,0) - 10,0 oreganize em sua 742ª Sessão Orde	intria malitada	Facuidade, o	Concurso de Titulos o de Drofessor Douto	e Provas para o	provimento de	Prof*	Dr* Marysia	Mara Rodrig	ues do Prado de Ca	rio: 10,0	Prof. Dr. Claud	io Miguel da Cost	a Neto: 10,0 -
am 24 de ju do pela Con	nho de 2008, homologou o Relatori nissão Julgadora.	io Final elabora-	de Dedicaç junto ao De	o Integral à Doctri partamento de Gené	cia e à Pesqui Itica - Disciplina	sa (R.D.I.D.P.), de Introdução	Prof. Prof*	Dr. Nilton M Dr ^a Anamari	azzer: 10,0 - a Siriani de l	9,2 - 9,5 - 9,6 Xiweira: 10,0 - 9,2 -	9,3 - 9,5	Prof. Dr. Edson Prof* Dr* Ragin	Antunes: 9,0 - 8,9 a Pokolmann Mark	- 9,0 - 9,5 us: 8,8 - 9,0 -
FACUL	DADE DE MEDICINA DE RIBEIRÃO P NICADO FMRP-USP Nº 047/2008	OT3RF	a Bioinform publicado n	stica, referente ao E o Diario Oficial do I	dital FMRP-USI Estado de 1º d	ri ^a 051/2007, le setembro de	ProP A Co	Dr ^a Eucenir A Ingregação A	redini Roch im sua 742	: 10,0 - 9,0 - 9,3 - Sessão Ordinária,	9,4 realizada	Candidata Vale Prof. Dr. William	ria Cristina Sandri Alves do Prado: 1	m: 9,5 - 9,5 - 8,5
Realiza Facuidado, o	ou-se nos días 12 e 13 de maio o o Concurso de Títulos e Provas para o	de 2008, nesta o provimento de	2007, cargo estavam in	número 1087070 / c scritos os candidat	daro número 10 os: Silvana Gi	01477, ao qual Iuliati, Ricardo	em 24 de do pela Co	junho de 20 omissão Julg	18, homolog adora.	ou o Relatório Final	elabora-	Prof* Dr* Maria - 9,2	Cristina de Oliveir	a Salgado: 9,0
01 (um) carg de Dedicaçã	go de Professor Doutor, referencia M ao Integral à Docencia e à Pesqui	IS-3, em Regime ilsa (R.D.I.D.P.),	Zarzeto Ni Antonio Ser	oteillo Vencio, Din pio Kimus Braz e Mô	io Rogerio Coi nica Guimarães	nete Franklin, Campiteli.	FACL	LDADE DE I UNICADO FI	MEDICINA D	E RIBEIRÃO PRETO 056/2008		Prof. Dr. Claudi Prof. Dr. Edson	o Miguel da Costa Antunes: 9,0 - 8,5	Neto: 9,0 -9,0 - 8,0 - 8,5
unto ao De 4207 - Bron	apartamento de Clínica Médica - D natología; RNM 4206 - Tecnología	de Alimentos e	Após a recimento a	instalação da Banca os candidatos Ricar	s, foi constatado Ido Zarzeto Nic	o não compa- coteillo Vencio,	Roal Faculdade	izou-se nos , o Concurso	dias 27 e 2 de Titulos e	28 de maio de 200 Provas para o provi	18, nesta mento de	Prof* Dr* Regin Candidata Nilci	a Pekelmann Mark Mitiko Matsuda:	aus: 9,0 - 9,0 -
RNM-4307 USP Nº 041	 Nutrição Experimental, referente a /2007, publicado no Diário Oficial d 	eo Edital FMRP- to Estado de O4	Antonio Ser A Corr	jio Kimus Braz e Mõ ssão Julgadora atribi	nica Guimarães uiu as seguintes	Campiteli. s notas à candi-	cargos de Dedicação	Professor D Integral & D	loutor, refer Jocéncia e a	encia MS-3, em Re Pesquisa (R.D.I.D.)	gime de P.), junto	Prof. Dr. William Prof* Dr* Maria	Alves do Prado: Cristina de Oliveir	8,0 - 7,5 - 0,0 a Salgado: 5,0
1000653, a	o qual estava inscrito o Professor	r Doutor Alceu	Examin	idor - Arguição do	Memorial - Pr	ova Didatica -	cirurgia	de Cabe	ça e Pe	gia, Utomnolaning scopo - Discipl	ina de	Prof. Dr. Claudi	o Miguel da Costa	Neto: 6.0 - 7,0
A Com	ao Junior. issão Julgadora atribuiu as seguinte	is notas ao can-	Prof* D	* Eucleia Primo Batic	ali Contel: 10,0	9,8 - 9,5 - 9,8	062/2007,	publicado	no Diário C	ficial do Estado d	e 04 de	Prof* Dr* Regin	a Pokolmann Mark	as: 7,0 - 6,8 -
Examin Examin	ador - Prova Didatia - Prova Escrita Modia Dondorada	a - Arguição do	Prof. D	Paulo Mazzoncini o	de Azevedo Mar	ques: 9.8 - 9,5	1087436/	claros núme	ros 420603.	1011839 e 100110	Bao qual	Prof. Dr. William	Alves do Prado: 1	9.5 - 10.0 - 9.5
Prof. Dr Prof [®] Dr	Ayrton Custodio Moreira: 10,0 - 10 * Selma Freire da Cunha: 10,0 - 10,0	0,0 - 10,0 - 10,0 0 - 10,0 - 10,0	Prof. D	Evandro Eduardo Se Renato Marcos End	aron Ruiz: 10,0 - Irizzi Sabbatini:	10,0 - 9,5 - 9,9 10,0 - 9,0 - 9,0	Hyppolito A Co	e Myriam de missão Julea	Lima Isaac. dora atribui	u as seguintes nota	s ao can-	10,0 - 10,0 Prof. Dr. Claud	io Miguel da Cost	a Neto: 10.0 -
Prof. Dr Prof. Dr	r. Sergio Zucoloto: 10,0 - 10,0 - 10,0 . José Eduardo Tanus dos Santos: 10	0 - 10,0 0,0 - 10,0 - 10,0	- 9,5 A Corr	ssão Julgadora atrib	uiu as seguinte	s notas ao can-	didato Mic Exami	guel Angelo inador - Pro	Hyppolito: ra Didatia -	Prova Escrita - Arg	sição do	9,7 Prof. Dr. Edson	Antunes: 9,5 - 9,5	- 9,5 - 9,5
10,0 Prof. Dr	r. Sergio Alberto Rupp de Paiva: 10,0	0 - 10,0 - 10,0 -	didato Dino Examin	kogerio Coinete Fran Idor - Argtiição do	nklin: Memorial - Pr	ova Didatica -	Memorial Prof [®]	 Media Pon Dr^a Wilma Te 	derada. rezinha Ans	elmo-Lima: 10,0 - 10	1.0 - 10.0	ProP Dr ^a Regin Candidato Wal	a Pékelmann Mark diceu Aparecido V	us: 9,5 - 10,0 arri Janior:
A Con	gregação em sua 742ª Sessão Orda	inária, realizada	Prova Escrit Prof ^e D	Média Ponderada Eucléia Primo Betic	ali Contel: 6,5 -	7,0 - 9,5 - 7,6	- 10,0 ProP	Dr ⁴ Maria Co	lia Cervi: 10	10 - 10,0 - 10,0 - 10	0	Prof. Dr. William Prof* Dr* Maria	Alves do Prado: Cristina de Oliveir	s,d - 10,0 - 8,0 a Salgado: 9,2
ini 24 de ju lo pela Con	nno ue 2008, nomologou o Relatori nissão Julgadora. DADE DE MEDICIPER DE DEPORTO TO D	io Final elabora- asciro	Prof D Prof. D	- Nilce Maria Martin Paulo Mazzoncini c	ez Hossi: 6,0 - 1 de Azevedo Mar	cu - 9.5 - 7.4 rques: 6,5 - 6,5	10,0 ProP	ur* Virginia ner suurur **	naes Leme I	Pernant: 10.0 - 10.0	- 10.0 -	Prof. Dr. Claudi	o Miguel da Costa	Neto: 8,5 - 9,5
COMU	NICADO FMRP-USP Nº 048/2008	do 2002	Prof. D	Evandro Eduardo S	eron Ruiz: 6.5 -	6.5 - 9.5 - 7.4	- 10.0	un i seittety Si	nita: 10.0	gnatart 10,0 - 10	.u - 10,0	ProP Dr ^a Regin	a Pekelmann Mark	us: 9,4 - 10,0
Faculdade, o 12 (deix) -	o Concurso de Titulos e Provas para o arrors de Professor Doutor oficial	o provimento de ncia MS-3 em	7,4 0. Pol	torio Final da Comi	isto kinater	concluie our	A Col	missão Julga	dora atribui	u as seguintes nota:	a candi-	Prof. Dr. William Prof. Dr. William	Alves do Prado: 1 Dristina de Oliveir	9,0 - 9,0 - 8,5 a Salandar - 9 6
RDLDP1	Dedicação Integral a Docencia junto ao Departamento de Cla	e a Pesquisa inica Médica -	diante dos da para res	isultados, a candida par o cargo de Pinfa	ta Silvana Giulia essor Doutor. re	ati seria indica- ferencia MS-3	Exami Memorial	inador - Pro - Média Pro	ra Didatia - derada.	Prova Escrita - Arg	piçao do	- 8.3 Prof. Dr. Claudi	o Miguel da Costa	Neto: 9.0 - 8.1
Disciplinas (de Estágio Supervisionado em Clin Estágio Supervisionado em Ambréo	ica Médica II - itorio de Climina	em RDIDP,	a Bioinformatica 4	to de Genética a Faculdada de	- Disciplina de Medicina de	Prof*1	Dr ^a Wilma Te	rezinha Ans	elmo-Lima: 10,0 - 10	1,0 - 10,0	Prof. Dr. Edson Prof [®] Dr ^a Ramin	Antunes: 8,5 - 8,0 Pekelmann Mart	- 80 - 82
Medica - RN I - RNM-44	4M-4407; Dietoterapia I - RNM-4402 10; Tópicos em Nutricão II - RNM-402	2 e Dietoterapia 306; Tópicos em	Ribeirto Pro	to da Universidade d pregação em sua 74	le Sao Paulo. 2ª Sessão Ontin	nária, realizada	Prof	Dr ^a Maria Ce Dr ^a Viroinia	lia Cervi: 10 Paes Leme	10 - 10,0 - 10,0 - 10 Ferriani: 10,0 - 10,0	10 - 10,0 -	O Relatorio F diante dos resultad	inal da Comissão os, o candidato I	Julgadora con sonardo Reve
Nutrição III Educação N	- RNM-4205; Exercicio da Profesa Autricional - RNM-4304 e Calvulve	io - RNM-4315; e Analises da	em 24 de ju do pela Cre	nho de 2008, hornok issão Julgadora.	ogou o Relatorio	Final elabora-	10.0 ProP	Dr ^a Shirley G	nizue Nacati	Pignatari: 10.0 - 10		Moraes seria indica de Professor Doute	do, por unanimida ar, referencia MS	ide, para ocup 3, em RDINO
Dietas para	Individuos - RNM-4303, referente a /2007, publicado no Diário Oficial d	ao Edital FMRP- to Estado de Na	COMI	NCADO FMRP-USP	DE REERROOP Nº 052/2008	ano -	- 10,0 Prof	Dr. Shiro Tre	nita: 10.0 -	10,0 - 10,0 - 10,0		Departamento de F Farmacológicas da	armacologia - Dis Terapéutica da Fa	ciplina RCG-2 culdade de M
de setembro claros núme	o de 2007, cargos nameros 108700 ros 1000861 e 1000888, ao gual es	02 e 1087010 / stavam inscrites	Realiz	u-se nos días 20 e Concurso de Titulos	21 de maio d e Provas para o	le 2008, nesta provimento de	A Co em 24 de	ingregação e junho de 20	im sua 742ª 38, homologi	Sessão Ordinaria, ou o Relatorio Final	realizada elabora-	Ribeirao Preto da U A Congregaria	niversidade de Sac o em sua 742º So	Paulo. ssão Ordinada
is Professor Diez Garcia	ras Doutoras Paula Garcia Chiarello	e Rosa Wanda	01 (um) can de Dedicac	o de Professor Douto lo Integral a Doctri	or, referència Mé icia e a Pescui	i-3, em Regime isa (R.D.I.D.P.)	do pela Co FACI	unissão Julg LOADE DE 1	adora. MEDICINA D	E RIBEIRÃO PRETO		em 24 de junho de : do pela Comissão à	2008, homologou Jigadora.	o Relatório Fin
A Com data Paula (issão Julgadora atribuiu as seguinte: Garcia Chirello:	s notas à candi-	junto ao De Obstetricia	Disciplina RCG-044	cologia e Obste 42, referente ac	tricia - Área de Edital FMRP-	COM Real	UNICADO FI Izou-se nos	dias 28 e 3	057/2008 29 de maio de 200	18, nesta	COMUNICADO	FMRP-USP Nº 06	0/2008
Examin Memorial - I	ador - Prova Didatia - Prova Escritz Media Ponderada.	a - Argüição do	USP nº 053 de setembr	2007, publicado no de 2007, cargo no	Diário Oficial de mero 1087177	> Estado de 06 / claro nomero	Faculdade 01 (um) ca	o Concurso Irgo de Profe	de Titulos e ssor Doutor,	Provas para o provis referência MS-3, en	nento de n Regime	Realizou-se n Faculdade, o Concu	os días 12 e 13 o so de Titulos e Pro	te junho de 2 vas para o pro
Prof. Dr Prof. Dr	r. Minton Cesar Foss: 10,0 - 10,0 - 10 r. Helio Vannucchi: 10,0 - 10,0 - 10,0	u.u - 10,0 0 - 10,0	1010794, a Maria Quin	o qual estava inscrita ana.	a Professora D	ioutora Silvana	de Dedica junto ao D	ição Integra lepartamente	a Docenci de Oftalmo	a e a Pesquisa (R. Ilogia, Otorrinolarin	u.I.D.P.), gologia e	un (um) cargo de Pri de Dedicação Integ	pressor Doutor, ref gral à Docència e	erencia MS-3, a Pesquisa (
Ptof. Dr	I. Kur Alberto Fernani: 9.5 - 10,0 - 10 Dapiel Fernani: 9.5 - 10,0 - 10	uu - 9.9	A Corr	ssao Juigadora atrib	uru as seguintes	notas a candi-	Disciplinas	e Cabeça (i de RFO-320	Hescoço - H - Sistema	wrea de Fonoaud e Funções Estomato	ulogia - graticas;	junto ao Departami Especial II - Patolog	mo de Patologia ja do Aparelho Re	 usciplinas di orodutor Masi
Prof. Dr	Contervenera da Conta. 100 - 10	0.0 - 10.0 - 10.0	Clara:	day Day 7	Descent Process	a second of the	DIC	- PC	data a	strained at 17.	Contract of the	Tanks 154		social Pro-

Notes: This is a sample of some layouts of job selection process postings in Brazilian government gazettes (similar to the Federal Register). Information on candidate scores, hiring committee members, exam types, selection process rules, results and other variables described in Figure 1 are contained in over 200 different layout types and can be found in the example inside green boxes. The underlying full data size spans over 35 million text documents published over almost 50 years.

FIGURE A.3: Examples of Raw Text Containing Hiring Information



Notes: This figure shows the implementation pipeline of the second step of the natural language processing algorithm developed in the paper to transform unstructured text into ready-to-use data. Each implementation step (on the left) is associated with a level of generated false positives and negatives (center) and underlying sample size (right).

FIGURE A.4: Algorithm Pipeline



Notes: This figure shows the gender share distribution of job applicants to various occupations and skill levels in Brazil's public sector from 1986 to 1990. Occupation titles in the data follow employer-specific career titles given each organizational structure. These titles are translated from Portuguese and then manually assigned occupation categories based on job or title description so that homogeneous occupation groups can be created. The occupation level displayed in the figure is intermediary — equivalent the Census Bureau Standard Occupational Classification (SOC) 4-digit code in most cases and slightly more granular in others. Skill levels are directly informed in job announcements, where only candidates attaining that educational level can apply for the job process. In the rare cases where different job titles are bridged by the same occupation name and they have distinct educational requirements, I consider the in the job title most closely reflecting the underlying occupational name or that is required more frequently. Occupations with blank bars had zero female applicants (e.g., carpenter, driver, mechanics) and some occupations had only women applying (e.g., data entry (support), spokesperson).

FIGURE A.5: Distribution of Female Applicants by Occupation and Skill Level



Notes: This figure compares aggregate statistics informed by Brazil's federal government on its public sector to calculated sample moments using data extracted from official government gazettes. For each statistic — annual number of job selection processes, number of hired employees, share of female hires, and share of high-skill jobs posted — the correlation between actual data and the constructed information using the NLP algorithm is above 99%, with error bands never outside 2%.

FIGURE A.6: NLP Algorithm Validation: Federal Jobs Sample Moments

59

			z = 0								
		Written	Non-written	Written & Non- written	Written Blind	Written Blind & Non- written					
	Written	Always Written									
	Non-written		Always Non- written								
z = 1	Written & Non-written			Always Written & Non-written							
	Written Blind	$w \rightarrow w(b)$ [20%]	nw ightarrow w(b) [6.7%]	$w + nw \rightarrow w(b)$ [6.7%]	Always Written Blind						
	Written Blind & Non- written	$w \to w(b) + nw$	$nw \rightarrow w(b) + nw$ [20%]	$w + nw \rightarrow w(b) + nw$ [46.7%]		Always Written Blind & Non- written					

Ω

Notes: This figure illustrates all possible potential treatments (strata) generated by the introduction of the impartiality requirement in hiring for federal jobs in Brazil. Areas shaded in gray are ruled out by standard differencein-differences assumptions and 5 out of the 6 allowed treatments are consistent with the variation induced by compliance: a job process transitioning from written exam to blind-written exam $(w \rightarrow w(b))$, a job process comprising a non-written exam switching to a blind-written $(nw \rightarrow w(b))$, or only adding the blind-written test $(nw \rightarrow w(b) + nw)$, and a hiring process using a mix of written and non-written tools potentially dropping the non-written and blinding the written $(w + nw \rightarrow w(b))$ or just blinding the written $(w + nw \rightarrow w(b) + nw)$. The potential treatment $w \rightarrow w(b) + nw$ is accounts for less than 1% of transitions in the data. Written exams are shorthand for written or multiple-choice tests, and non-written indicate interviews, practical exams, or oral exams. Numbers in $[\cdot]$ give the frequency of each treatment type in the estimating sample.

FIGURE A.6: Potential Treatment Space Generated by the Impartiality Requirement

Treatment	Occupation
$w \longrightarrow w(b)$	translator, librarian, low-level bureaucrat, school officer, office staff, community healthcare worker
$nw \longrightarrow w(b)$ $nw \longrightarrow w(b) + nw$	accountant, psychologist, telephonist, social worker, truck driver nurse, pharmacist, receptionist, typist, visual designer, chemist
$w + nw \longrightarrow w(b)$	kitchen assistant, nutritionist, cleaner, security agent, courier, civil engineer, programmer
$w + nw \longrightarrow w(b) + nw$	teacher, judge, mason, cook, professor, high-level bureaucrat, veterinarian, police officer

TABLE A.1: Examples of Occupations in Each Treatment Type

Notes: These are examples of occupation titles in each treatment type generated by the introduction of impartiality in hiring in Brazil's public sector.

Entity	Online Archives Available Since	Government Level
Brazil	1808	Federal
Rondônia	2011	State
Acre	2010	State
Amazonas	1956	State
Roraima	1998	State
Pará	2016	State
Amapá	1988	State
Tocantins	2005	State
Maranhão	2001	State
Piauí	2004	State
Ceará	1999	State
Rio Grande do Norte	—	State
Paraíba	2003	State
Pernambuco	1936	State
Alagoas	2010	State
Sergipe	2012	State
Bahia	2007	State
Minas Gerais	2011	State
Espírito Santo	2006	State
Rio de Janeiro	2005	State
São Paulo	1891	State
Paraná	2004	State
Santa Catarina	2011	State
Rio Grande do Sul	1968	State
Mato Grosso do Sul	1979	State
Mato Grosso	1967	State
Goiás	2008	State
Distrito Federal	1967	State

TABLE A.2: Raw Text Data Availability: Government Official Gazettes

Notes. This table shows the primary sources of job hiring processes in various levels in Brazil's public sector. Each administrative level displayed publishes its own official gazette in a separate online repository. The middle column lists dates when online archives of each journal became available.

	Log # Candidates				
	All (1)	Women (2)	Men (3)		
$\operatorname{Federal}_j \times \operatorname{Post}_{j(t)}$	-0.245 (0.322)	-0.212 (0.382)	-0.316 (0.251)		
Occupation FE	Х	Х	Х		
Year FE	Х	Х	Х		
Obs.	86,959	86,959	86,959		

TABLE A.3: Estimates of Screening Impartiality on Job Process Outcomes

Notes: The table shows difference-in-differences estimates of the log number of job applicants using the regression:

$$\log (\# \text{Candidates}_{jt}) = \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_{o(j)} + \theta_t + u_{ujt}$$

where $\text{Post}_{j(t)}$ is a dummy equal to 1 if a job process occurred $t \ge 1989$ and $\text{Federal}_j = 1$ if it was for a federal job. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

		Final Score							
	Wo	men	Men						
	(1)	(2)	(4)						
$\text{Federal}_{j} \times \text{Post}_{j(t)}$	0.067**	0.176***	-0.075^{*}	-0.233***					
	(0.030)	(0.018)	(0.037)	(0.092)					
Applicant FE		Х		Х					
Occupation FE	Х	Х	Х	Х					
Year FE	Х	Х	Х	Х					
Obs.	54,892	10,324	32,067	7,825					

TABLE A.4: Estimates of Screening Impartiality on Candidate Scores, Candidate FE

Notes: The table shows difference-in-differences estimates of outcomes for applicant *i* in job process *j* in a given year *t*. The first specification is $y_{ijt} = \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_{j(o)} + \theta_t + u_{ijt}$ only with female candidates in column (1) and only with male candidates in column (3). The second specification includes an additional job applicant fixed effect, in columns (2) and (4). The outcome *y* represents either a candidate's final score, written score (written exams), or non-written score (interview, oral, practical exams). If a job process occurred $t \ge 1989$, $\text{Post}_{j(t)}$ is a dummy equal to 1, and if it was for a federal job, $\text{Federal}_j = 1$. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

		Final Score						Relative Rank	Relative Score
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
$\operatorname{Federal}_{j} \times \operatorname{Post}_{j(t)}$	0.141**	0.141***	0.141**	0.127***	0.127***	0.139***	0.140***	-0.025***	0.038**
\times Female _i	(0.064)	(0.017)	(0.064)	(0.050)	(0.050)	(0.055)	(0.050)	(0.011)	(0.018)
Clustering									
Employer	X								
Government level		X							
Occupation			X						
Job process				Χ	Χ	X	Χ	X	X
Fixed effects									
Occupation code	X	X	X					X	X
Occupation skill				X					
Census region					X				
Occupation code×Year						X			
Year	X	X	X	X	Χ		X	X	X
Controls									
Applicant pool size							X		

TABLE A.5: Effects of Hiring Impartiality on Gender Gaps: Alternative Specifications and Robustness

Notes: The table reports alternative specifications of the baseline triple-difference model in the paper: $y_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(0)} + \theta_t + u_{ijt}$. Columns (1) through (3) maintain this specification but alternatively cluster standard errors by employer codes, government (at the treatment level), or occupation codes, instead of the baseline clustering by job processes. Column (4) replaces the occupation code fixed effect $\delta_{j(0)}$ with a set of occupational skill dummies (less than high school high school, less than college, college), column (5) uses Brazilian census regions dummies (North, Northeast, South, Southeast, West), and column (7) uses the interaction $\delta_{j(0)} \times \theta_t$ as fixed effect. Column (7) runs the baseline model but includes the applicant size pool in each job process as control. Column (8) also runs the baseline specification, but uses a job applicant's relative rank in the selection process instead of the standardized final score. A relative rank is the candidate's position in proportion to the total number of ranked applicants, such that the best-ranked candidate is measured as $\frac{1}{\# \text{ candidates}}$. Column (9) uses a relative final score as outcome, measured as the candidate's score divided by the highest score in that job process, such that the best-ranked applicant is normalized to 1.

		Final	Score	$\mathbb{P}(Hired Applied = 1)$			
	<high School</high 	High School	High College or school Advanced Degree		High School	College or Advanced Degree	
	(1)	(2)	(3)	(4)	(5)	(6)	
$\begin{aligned} & \text{Federal}_j \times \text{Post}_{j(t)} \\ & \times \text{Female}_i \end{aligned}$	-0.016 (0.084)	0.160 (0.111)	0.204*** (0.080)	0.002 (0.003)	-0.001 (0.002)	0.011* (0.005)	
Occupation FE	Х	Х	Х	Х	Х	Х	
Year FE	Х	Х	Х	Х	Х	Х	
Obs.	35,475	22,071	29,413	35,475	22,071	29,413	

TABLE A.6: Estimates of Effects on Scores and Hiring Probability, Skill Level

Notes: This table reports triple-difference estimates of the model $y_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(o)} + \theta_t + u_{ijt}$, with final scores as the outcome in columns (1) through (3) and a dummy equal to one if the candidate is hired for columns (4) to (6). Within each outcome, columns represent regressions in subsamples based on the education required by a job process from applicants. If a job process occurred $t \ge 1989$, $\text{Post}_{j(t)}$ is a dummy equal to 1, and if it was for a federal job, Federal_j = 1. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

	Final Score			$\mathbb{P}(Hired \text{Applied} = 1)$		
Occupation Gender Identity	Female	Neutral	Male	Female	Neutral	Male
A. All Job Applicants	(A.1)	(A.2)	(A.3)	(A.4)	(A.5)	(A.6)
$\begin{aligned} & \text{Federal}_j \times \text{Post}_{j(t)} \\ & \times \text{Female}_i \end{aligned}$	0.305*** (0.105)	0.149*** (0.043)	0.267** (0.117)	0.012* (0.006)	0.005* (0.002)	0.002** (0.001)
B. Female Job Applicants	(B.1)	(B.2)	(B.3)	(B.4)	(B.5)	(B.6)
$\operatorname{Federal}_{j} \times \operatorname{Post}_{j(t)}$	0.066*** (0.019)	0.061** (0.026)	0.151* (0.091)	0.003** (0.001)	0.0007 (0.001)	-0.0005 (0.001)
C. Male Job Applicants	(C.1)	(C.2)	(C.3)	(C.4)	(C.5)	(C.6)
$\operatorname{Federal}_{j} \times \operatorname{Post}_{j(t)}$	-0.238*** (0.089)	-0.087** (0.038)	-0.115*** (0.035)	-0.008* (0.005)	-0.004** (0.002)	-0.002*** (0.001)
Occupation FE	Х	Х	Х	Х	Х	Х
Year FE	Х	Х	Х	Х	Х	Х
Obs.	48,681	16,848	21,430	48,681	16,848	21,430

TABLE A.7: Estimates on Scores and Hiring Probability, Occupation Feminization

Notes: This table displays regression coefficients of the triple-differences specification $y_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(o)} + \theta_t + u_{ijt}$ in Panel A, and of the difference-in-differences model $y_{ijt} = \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_{j(o)} + \theta_t + u_{ijt}$ only with female applicants in Panel B and male applicants in Panel C. The outcome represents either a female candidate's final score relative to a male candidate in the first three columns, or the probability that a female candidate receives a job offer relative to a male candidate in the last three columns. If a job process occurred $t \ge 1989$, $\text{Post}_{j(t)}$ is a dummy equal to 1, and if it was for a federal job, $\text{Federal}_j = 1$. Columns (1) and (3) report estimates for a subsample of female-dominated occupations, defined as the proportion of women in that occupation > 60%. Columns (2) and (5) run the regression for a subsample of occupations that are neutral or gender balanced, if the proportion of women in that occupation is between 40% and 60%. Columns (3) and (6) run the regression for a subsample of male-dominated occupations, defined as the share of women < 40%. Standard errors are clustered at the job process level. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

	$\mathbb{P}(nw \longrightarrow w(b) + nw pre \in nw)$ (1)	$\mathbb{P}(w + nw \longrightarrow w(b) + nw pre \in w + nw)$ (2)
High-skill	0.183	0.276*
	(0.155)	(0.152)
Female-dominated	0.061	0.068
	(0.151)	(0.150)
% female applicant	0.718	0.771
	(0.611)	(0.628)
Candidates/openings	-0.002	-0.001
	(0.004)	(0.003)
<i>R</i> -squared	0.06	0.09
Occupations	52	47

TABLE A.8: Correlation Between Occupation Characteristics and Selection into Treatment

Notes: This table shows regressions at the occupation-level determining the relationship between the probability that an occupation follows one of two possible treatments and its characteristics. The first column regresses occupations that only used an interview before the impartiality requirement (nw) on the probability of switching to w(b) + nw instead of w(b). Column (2) regresses the probability that occupations using written and non-written exams before the time of treatment (w + nw) receive treatment w(b) + nw as opposed to w(b).
	Final Score				$\mathbb{P}(Hired = 1)$			
Pre-treatment mix:	nw		w + nw		nw		w + nw	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Overidentification test <i>p</i> -value	0.86	0.36	0.91	0.75	0.14	0.17	0.00	0.61
χ^2 statistic	3.9	8.7	10.7	15.4	12.2	11.5	122.2	18.66
Female applicant	X		X		X		X	
Male applicant		X		X		X		X
Occupation FE	Х	Х	Х	Х	Х	Х	X	X
Year FE	X	X	X	X	X	X	X	Х

TABLE A.9: Selection Tests for Multiple Treatments

Notes: This table reports the results of overidentification tests for the multiple treatments generated by the impartiality requirement in hiring. The reported Sargan (1958) statistics compare whether just-identified estimates using alternative job titles as instruments for a treatment are equal under the null hypothesis. Columns (1), (2), (5), and (6) include treatments $nw \longrightarrow \{w(b), w(b) + nw\}$, while the remaining columns include treatments $w + nw \longrightarrow \{w(b), w(b) + nw\}$. The tests are generated with the model

$$y_{ijt} = \kappa_2 \operatorname{Treatment}_{j(d)} + \kappa_3 \left(\operatorname{Post}_{j(t)} \times \operatorname{Treatment}_{j(d)} \right) + \delta_{o(j)} + \theta_t + u_{ijt}$$

where $\text{Treatment}_{j(d)}$ replaces the $Federal_j$ dummy, mapping to each possible treatment conditional on the pre-existing mix, and equal to 0 for state jobs. Regressions are ran separately for men and women. The outcomes are a job applicant final score and probability of being hired.

A. Gender Final Score Gap Gap					
	$w \longrightarrow w(b)$ (A1)	$nw \longrightarrow w(b)$ (A2)	$nw \longrightarrow nw + w(b)$ (A3)	$w + nw \longrightarrow w(b)$ (A4)	$w + nw \longrightarrow nw + w(b)$ (A5)
$\text{Federal}_j \times \text{Post}_{j(t)}$	0.040***	0.195***	0.141***	0.026*	0.002
\times Female _i	(0.016)	(0.044)	(0.016)	(0.014)	(0.025)
B. Gender Hiring Gap					
	$w \longrightarrow w(b)$	$nw \longrightarrow w(b)$	$nw \longrightarrow nw + w(b)$	$w + nw \longrightarrow w(b)$	$w + nw \longrightarrow nw + w(b)$
	(B1)	(B2)	(B3)	(B4)	(B5)
$\text{Federal}_j \times \text{Post}_{j(t)}$	0.005**	0.070***	0.058***	0.005	-0.005
\times Female _i	(0.002)	(0.013)	(0.007)	(0.003)	(0.007)
Occupation FE	X	X	X	X	X
Year FE	Х	Х	Х	Х	Х
Obs.	12,066	9,343	18,570	44,964	32,898

TABLE A.10: Treatment Effects on Gender Gaps of Changes in Screening Tools

Notes: This table shows treatment effects estimated from the triple-difference model:

$$y_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right)$$

+ $\delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right)$
+ $\beta_d \left(\text{Federal}_j \times \text{Post}_{t(j)} \times \text{Female}_i \right) + \delta_{o(j)} + \theta_t + u_{ijk}$

which compares outcomes (y_{ijt}) for female candidates relative to men (Female_{*i*}) participating in job processes for the same occupation $(\delta_{\sigma(j)})$ that had screening practices changed following treatment *d*. Treatments are displayed at the top with their corresponding estimates in the bars displayed below. Bars are 95% confidence intervals obtained with standard errors clustered at the job process level.

Treatment Group (<i>d</i>): $w + nw \longrightarrow w(b) + nw$										
		Final Score					Non-Blind Score		$\underline{\mathbb{P}(Hired \text{Applied}=1)}$	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	
$\begin{aligned} & \text{Federal}_j \times \text{Post}_{j(t)} \\ & \times \text{Female}_i \end{aligned}$	0.009 (0.014)	0.052*** (0.022)	-0.011 (0.042)	0.059*** (0.017)	-0.039 (0.040)	0.011 (0.008)	-0.006 (0.009)	0.014*** (0.005)	0.009 (0.026)	
Blind Score				0.630*** (0.049)	0.446*** (0.029)					
Job Process Blind Weight		> 50%	< 50%	> 50%	< 50%	> 50%	< 50%	> 50%	< 50%	
Occupation FE Year FE	X X	X X	X X	X X	X X	X X	X X	X X	X X	

TABLE A.11: Decomposing Treatment Effects by Weight of Blind Stages

Notes: This table plots treatment effects for treatment type *d* where jobs process transition from using a mix of written and non-written tools to blinding the written $(w + nw \rightarrow w(b) + nw)$. The model implemented is the triple-differences specification $y_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_i \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(o)} + \theta_t + u_{ijt}$. If a job process occurred $t \ge 1989$, $\text{Post}_{j(t)}$ is a dummy equal to 1, and if it was for a federal job, $\text{Federal}_j = 1$. The outcome *y* represent either a candidate's final score, non-blind (non-written) score, or probability of being offered a job. Columns (2), (4), (6) and (8) condition on the weight on the blind written test in a job process to be > 50%, and columns (3), (5), (7) and (9) for the weight on the blind written test to be less than 50%. Standard errors are clustered at the job process level.

	$\mathbb{P}(Hired \text{Applied} = 1)$							
	$\begin{array}{c} w \longrightarrow w(b) \\ (1) \end{array}$	$nw \longrightarrow w(b)$ (2)	$nw \longrightarrow nw + w(b)$ (3)	$w + nw \longrightarrow w(b)$ (4)	$w + nw \longrightarrow nw + w(b)$ (5)			
$\begin{aligned} & \text{Federal}_j \times \text{Post}_{j(t)} \\ & \times \text{Female}_i \end{aligned}$	0.005**	0.070***	0.058***	0.003	-0.004			
	(0.002)	(0.013)	(0.007)	(0.004)	(0.008)			
Occupation FE	X	X	X	X	X			
Year FE	X	X	X	X	X			
Occupation Skill FE	X	X	X	X	X			
Obs.	12,066	9,343	18,570	44,964	32,898			

TABLE A.12: Treatment Effects of Changes in Screening Tools: Skill Robustness

Notes: This table shows treatment effects for each treatment type *d* implementing the tripledifference specification $\mathbb{P}(Hired = 1)_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(o)} + \theta_t + u_{ijt}$, with an included set of fixed effects accounting for occupation skill. If a job process occurred $t \ge 1989$, $\text{Post}_{j(t)}$ is a dummy equal to 1, and if it was for a federal job, $\text{Federal}_j = 1$. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

	$\mathbb{P}(Hired \text{Applied} = 1)$							
	$w \longrightarrow w(b)$ (1)	$nw \longrightarrow w(b)$ (2)	$nw \longrightarrow nw + w(b)$ (3)	$w + nw \longrightarrow w(b)$ (4)	$w + nw \longrightarrow nw + w(b)$ (5)			
$\begin{aligned} & \text{Federal}_j \times \text{Post}_{j(t)} \\ & \times \text{Female}_i \end{aligned}$	0.005** (0.002)	0.070*** (0.013)	0.057*** (0.007)	0.005 (0.003)	-0.005 (0.007)			
Occupation FE Year FE	X X X	X X X	X X X	X X X	X X X			
Obs.	12,066	× 9,343	18,570	× 44,964	32,898			

TABLE A.13: Treatment Effects of Changes in Screening Tools: Feminization Robustness

Notes: This table shows treatment effects for each treatment type *d* implementing the tripledifference specification $\mathbb{P}(Hired = 1)_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(o)} + \theta_t + u_{ijt}$, with an included set of fixed effects accounting for the degree of feminization of the occupation (female or male dominated, with neutral as the omitted group). If a job process occurred $t \ge 1989$, $\text{Post}_{j(t)}$ is a dummy equal to 1, and if it was for a federal job, $\text{Federal}_j = 1$. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

	$\mathbb{P}(Hired \text{Applied} = 1)$						
	$w \longrightarrow w(b)$	$nw \longrightarrow w(b)$	$nw \longrightarrow nw + w(b)$	$w + nw \longrightarrow w(b)$	$w + nw \longrightarrow nw + w(b)$		
Panel A. Less Than College	(A.1)			(A.4)	(A.5)		
$\operatorname{Federal}_j \times \operatorname{Post}_{j(t)}$	0.004***			0.012*	0.006		
\times Female _i	(0.001)			(0.006)	(0.011)		
Panel B. College or More	(B.1)	(B.2)	(B.3)	(B.4)	(B.5)		
$\operatorname{Federal}_{j} \times \operatorname{Post}_{j(t)}$	0.005*	0.085***	0.055***	-0.006	-0.017		
\times Female _i	(0.002)	(0.020)	(0.006)	(0.006)	(0.011)		
Occupation FE	X	Х	X	X	X		
Year FE	Х	Х	Х	Х	Х		

TABLE A.14: Treatment Effects of Changes in Screening Tools by Occupation Skill

Notes: This table shows treatment effects for each treatment type d implementing the tripledifference specification $\mathbb{P}(Hired = 1)_{ijt} = \delta_1 \text{Female}_i + \delta_2 \text{Federal}_j + \delta_3 \left(\text{Post}_{j(t)} \times \text{Federal}_j \right) + \delta_4 \left(\text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_5 \left(\text{Federal}_j \times \text{Female}_i \right) + \beta \left(\text{Federal}_j \times \text{Post}_{j(t)} \times \text{Female}_i \right) + \delta_{j(o)} + \theta_t + u_{ijt}.$ If a job process occurred $t \ge 1989$, $\text{Post}_{j(t)}$ is a dummy equal to 1, and if it was for a federal job, $\text{Federal}_j = 1$. Panel A uses a subsample of candidates applying to jobs requiring less than a college degree, while Panel B estimates off job applicants to college positions. All specifications contain year and occupation fixed effects and have standard errors are clustered at the job process level.

Appendix B Transforming Unstructured Text Into Data

To organize ideas, consider the following sequence of tasks necessary to automate the construction of a large-scale applicant-performance :

- 1. Filter out all text contained in official government documents unrelated to hiring steps
- 2. Define the boundaries of the relevant text
- 3. Identify the underlying job process of a certain relevant text
- 4. Link different postings belonging to the same process.
- 5. Transform text in each posting into data.

Due to the layout of Brazilian official journals, each step above presents a host of issues. First, there are no boundaries between the text of a job posting and other information — say another job posting or a list of government contractors suspended — so that defining the domain of relevant information ex-ante is difficult. Then, because surrounding text may be of a similar nature, filtering out extraneous information that does not belong to a specific job process is also challenging. Further complicating matters, the different stages of the same job process have no exclusive identifier (e.g., a hiring process code) and subsequent postings rarely mention the date that the *Edital* (job announcement) was published. Taken together, these issues underscore the limitations of relying on any text-selection method based on existing content structure to automate steps 1 through 4.

Given that one could identify and link the precise text domain of all stages of a hiring process, extracting data from the raw text presents an even bigger challenge. There is no pre-determined layout or set of rules instructing how postings in the *Diários* should display information. Some postings may present candidate results in tables, others in continuous text; scores may be organized by exam type or committee member, or a combination of both; exam types are sometimes informed near candidates and scores and other times at the beginning of the journal posting. While there is certainly some commonality across official postings, after all, these have legal content and enforcement and are often submitted by specialized bureaucrats on behalf of the employer, these similarities are subtle and offer little aid to scrape-like tools that rely on well-defined patterns.

A Conceptual Implementation

While all steps of hiring processes in the Brazilian public sector are carefully documented and publicly available, there are two major challenges to systematically using these raw data sources. The first is that published notices within the same hiring process are not directly linked. In practice, it is non-trivial to assign a list of candidate scores posted in a certain journal issue to a previously-published job announcement information. Off-the-shelf text analysis tools that connect text bodies based on proportionality and similarity, like the term frequency-inverse document frequency (tf-idf) and cosine similarity, are not useful in this context since information in legal publications is highly confounding. The same page of an official gazette might contain sections with a hiring round of eye surgeons at a certain hospital and a section with another job selection process of brain surgeons at the same hospital. In other cases, the same hospital might be hiring eye surgeons through more than one public notice.

A.1 Defining Textual Matching Attributes

Conceptually, the problem of linking steps of the same job process boils down to connecting a number of T text snippets by matching on N text attributes. Both T and N are ex-ante unknown. A job selection process might

have any number *T* of published texts and it is unclear which and how many *N* lexical structures one might need to properly connect such announcements. How should *N* be chosen? Consider that a sequence of t = 1, ..., T connected text documents can be summarized by the set of attributes A^t :

 $A^{t} = \{$ message keyword, sender, release date, message keyword feature $\}$

In the case of a specific hiring process, these attributes take the correspondence

{job, employer, release date, job feature}

where job feature might refer to the place of work, position title, or any dimension that distinguishes A^t from A^j given $A^t \setminus \{job attribute\} = A^j \setminus \{job attribute\}, t \neq j$. The motivation for defining A^t stems from its search-query use. For each government gazette issue, I search for a job posting notice, using a combination of words in the same paragraph (formally defined as some text string neighborhood) comprised of "announcement", "job", "hiring", and "posting". When there is one or more hits, I bound the relevant text to each job announcement and extract attributes A (the implementation of relevant text boundaries is detailed below). Only the *release date* is ex-ante known, since I know when each journal issue is published. To correctly identify the terms containing the other attributes in A, I rely on ad hoc dictionaries and allow them to expand by "learning" new terms.

More precisely, I construct a list with all public entities from government webpages and a dictionary of occupations that provide a fairly broad library to search for full or partial matches in job announcement texts. After I identify a job (message keyword) and employer (sender) pair, I update these dictionaries used in the search query for the same keyword. For example, my initial occupation library contains "Professor" and adds terms like "Assistant Professor", "Associate Professor", and so on as I progressively incorporate richer versions of the message keyword "Professor". After building the set of attributes that uniquely identifies a job hiring process, I search in all documents published after the release date for occurrences of A^t . The collection of T text excerpts containing A^t thus comprises all published notices of the job selection process.³⁴

Suppose a researcher wants to use the New York Times online archives to collect data on murder rates in major US cities since 1890. In this case, the message keyword could be "murder rate", a list with the desired city names would inform different values for the sender, the release date is the issue's date, and the message keyword feature could be a year matching the release date. Instead of going through multiple manual searches in the archived texts for each combination of city and year, the results to the approach above would give the relevant text snippets for the next stage: transforming the text into data.

A.2 Data Extraction

After linking hiring rounds across government gazette issues, the next challenge is the lack of structure in the published notices. Hiring rounds might be displayed in tables of varying dimensions, in free text, or in a combination of both. In most text analysis applications (e.g. Atalay et al. (2020)) every text snippet has a fairly similar structure, which greatly facilitates mining. Moreover, even in cases with free text as in Bybee et al. (2020), the underlying text structure is relevant only to the extent that it conveys information to identify a predictor based on

³⁴Note that while still relying on some dictionaries to discipline the domain of the message keyword and sender types, this approach takes an agnostic view with respect to the information derived from the underlying text contents and its potential use to connect text snippets, as well as the need for computationally-intensive updating of the initial search libraries. Indeed, in most applications, researchers may not even need to update their initial search parameters.

the message content. That is, researchers map text (raw or represented by a numerical array) onto a discrete set of measures $\mathcal{T} \rightarrow \{M_1(\tau), M_2(\tau), \dots, M_K(\tau)\}$, where τ is a transformation of the underlying raw text. Such mappings include sentiment-based approaches as in Gentzkow et al. (2019), where the true sentiment of a message is transformed into a function of a latent quantity.

In many applications, however, researchers might be interested in extracting exact information from text and converting that into a database by distilling \mathcal{T} into a pre-determined list of variables $\{x_1, x_2, \dots, x_K\}$. This is usually an extremely time-intensive task, highly dependent on the particular context and that relies heavily on strong prior information about the potential variations of text structure across \mathcal{T} . To solve this issue, I leverage the semantic structure implied by the relationship between listed variables, so that a fixed variable x_1 conditions all other data points that a researcher is interested in extracting, $\{x_1, x_2 | x_1, \dots, x_K | x_1\}$.

To see how, let x_1^i correspond to candidate *i*'s exam score, x_2^i her name, x_3^i the exam type and x_4^i the committee member who gave score x_1^i . In order to deal with the unstructured nature of the text, I start by targeting text tokens containing numbers, which is the only morphology that maps onto exam scores. Of course, many numbers within the text might be extraneous and not represent scores. The next step searches for tokens in the neighborhood of every number that match the characteristics of each additional variable *x*. This both fully defines the other variables that relate to x_1 and filters out numeric elements that are not scores.³⁵

Often times the implementation of an automated tool to extract data in these cases is so burdensome that researchers end up hand-collecting the desired variables from a feasible subsample of text documents. But in many cases this is simply not feasible. By choosing one variable to which most or all of the other desired variables relate, this approach avoids reliance on information from semantic structure that differs across public announcements, and focuses on relationships that organize each candidate's relevant information in the same way within a job notice text. The underlying semantic structure thereby informs the selection model about the location of certain variables rather than feed a label grouping, such as political slant or favorability of a review. This step requires the use of few *ad hoc* dictionaries (a list with Brazilian names in the current application and another with different examination types), which are allowed to learn similarly to before with the lists of occupations and employer names.

Returning to the application example of historical murder rates in major US cities, after defining the relevant NYT articles containing murder rates of a city in a given year (step 1), now the researcher implements step 2 to extract the actual number from the text (x_1), which is the murder rate. The process here is simple since *i*) the murder rate number only has two relevant attributes — city and period or year. Of course, numeric values of x_1 may give different scales or measurements of murder figures, for which the researcher will need to implement some form of ex-post harmonization.

³⁵For instance, numbers without recognizable names in their vicinity are discarded. Further, the same candidate might have several scores for different exams, which will differ along some dimension (Exam I and Exam 2, Written Exam and Oral Exam, etc.). This attribute will be relevant not only for individual *i*'s score, but also for all other candidates who took the same exam type. Thus, it must be that the relation between x_1^i and x_3^i holds for all $i \neq g$. For example, if the data is organized in a table where a certain column contains each exam type and rows display candidate names and scores, each candidate's score in a given exam will be aligned with the column's name. Another example: if the beginning of recorded scores displays a legend that gives an ordering such as "*Name - ID # - Written Exam - Interview - Final Score - Rank*", every candidate will have scores displayed in the same order.

Appendix C Model Solution

This Appendix shows the complete solution to the model described in Section 5. The environment and structure of hiring decision are as described there.

A Hiring Rates

A.1 Hiring Rate With Written Exams

When the hiring technology only includes written tests, the distribution of written signals, $s^* = y + v_s(x) + \varepsilon_s$, $\varepsilon_s \sim N(0, 1/h_s)$, is given by:

$$s^{*} \sim N\left(y + \nu_{s}\left(x\right), 1/h_{s}\right)$$

where *s* represents the unbiased signal $s = y + \varepsilon_s$, h_s is the inverse of the variance of the written signal, measuring the precision of written testing and independent of group membership *x*. After observing *s*^{*}, the hiring manager updates her assessment of expected productivity of candidates, initially based on group productivity, $\mu_0(x)$, forming the posterior:

$$\mu(x,s^*) = s \frac{h_s}{h_0 + h_s} + \mu_0(x) \frac{h_0}{h_0 + h_s} + \nu_s(x).$$

Given the written signal, *s*, and the perceived group productivity, $\mu_0(x)$, the evaluator updates her assessment of expected productivity of candidates according to:

$$y|_{s} \sim N(\mu(x,s), 1/(h_{0}+h_{s})).$$

Here, the updated degree of precision is $(h_0 + h_s)$ and the updated mean equals:

$$\mu(x,s) = s \frac{h_s}{h_0 + h_s} + \mu_0(x) \frac{h_0}{h_0 + h_s} + \nu_s(x).$$

The hiring decision that maximizes the evaluator's objective function satisfies the rule Hire = $I{\{\mu(x,s) > k_s\}}$, where k_s is the screening threshold that yields a fixed hiring rate of $K \in (0, 0.5)$. Plugging the expression for $\mu(x,s)$ into the hiring rule yields the following:

$$s > \frac{(h_0 + h_s)(k_s - \nu_s(x) - d_s \pi_j(x)) - h_0 \mu_0(x)}{h_s}.$$

Since the distribution of *s* is $N(\mu_0(x), 1/h_0 + 1/h_s)$, the above inequality can be rewritten as:

$$\frac{s - \mu_0(x)}{\frac{1}{h_0} + \frac{1}{h_s}} > \frac{(h_0 + h_s)(k_s - \nu_s(x) - d_s\pi_j(x) - \mu_0(x))}{h_s\sqrt{(\frac{1}{h_0} + \frac{1}{h_s})}}$$

which can finally be expressed as:

$$\frac{s - \mu_0(x)}{\frac{1}{h_0} + \frac{1}{h_s}} > \underbrace{\frac{k_s - \nu_s(x) - \mu_0(x) - d_s \pi_j(x)}{\sigma_0 \rho_s}}_{Z_s^*(x)}$$
(C.1)

where $\rho_s \equiv Corr(\mu(x,s), y) = (1 - \frac{h_0}{h_0 + h_s})^{1/2}$ and $z_s^*(x)$ is the hiring threshold for group x determined by using written exams. The expected hiring rate of applicants from group x is $1 - \Phi(z_s^*(x))$.

A.2 Hiring Rate With Non-Written Exam

When the employer screens job applicants solely based on non-written tests, the intuition for the effect of evaluator bias, precision, and tool bias is similar to the case of written tests. However, an important distinction arises as a consequence of different subjectivity degrees between the two practices. Formally, let the distribution of nonwritten signals be $\eta^* = y + \nu_{\eta}(x) + \varepsilon_{\eta}$, $\varepsilon \sim N(0, 1/h_{\eta})$, where $\nu_{\eta}(x)$ represents the possible disparate impact of non-written tests and η is the unbiased non-written signal, $\eta = y + \varepsilon_{\eta}$. Non-written exams allow discretion d_{η} to evaluators. Given that interviews or oral exams are more subjective than written tests, the discretion given to the evaluator is higher with non-written than with written tests: $d_{\eta} > d_s$.

The hiring rate for group x when candidates are screened using non-written tests is obtained following the same steps as in the previous case, observing the different distribution of non-written signals. In this case, an applicant screened with a non-written exam is hired if

$$\frac{\eta - \mu_0(x)}{\frac{1}{h_0} + \frac{1}{h_\eta}} > \underbrace{\frac{k_\eta - \nu_\eta(x) - \mu_0(x) - d_\eta \pi_j(x)}{\sigma_0 \rho_\eta}}_{z_{\pi}^*(x)}$$
(C.2)

The corresponding probability that a group *x* candidate is hired is given by $1 - \Phi(z_n^*(x))$.

A.3 Hiring Rate With Written and Non-Written Exams

Given the two signals previously determined, η^* and s^* , and the perceived group productivity, $\mu_0(x)$, the evaluator updates her assessment of expected productivity according to:

$$y|_{\eta^*,s^*} \sim N(\mu(x,\eta^*,s^*),1/(h_0+h_\eta+h_s)).$$

From the above, the updated degree of screening precision is $h_0 + h_\eta + h_s \equiv h_T$ and the updated posterior is thereby:

$$\mu(x,\eta^*,s^*) = s\frac{h_s}{h_T} + \eta\frac{h_\eta}{h_T} + \mu_0(x)\frac{h_0}{h_T} + \nu_s(x)\frac{h_s}{h_T} + \nu_\eta(x)\frac{h_0 + h_\eta}{h_T}$$

Thus, the hiring decision is given by:

$$\mu(x,\eta,s) > k_T - \pi(x)(d_\eta + d_s)$$

$$\frac{h_s s + h_\eta \eta + h_0 \mu_0(x)}{h_T} > k_T - \pi(x)(d_\eta + d_s) - \nu_s(x)\frac{h_s}{h_T} - \nu_\eta(x)\frac{(h_0 + h_s)}{h_T}.$$

Since $\eta = y + \varepsilon_{\eta}$, $s = y + \varepsilon_s$, and $y, \varepsilon_{\eta}, \varepsilon_s$ are independent, the left-hand side of the above inequality is distributed as:

$$\begin{split} \frac{h_s s + h_\eta \eta + h_0 \mu_0(x)}{h_T} &\sim N\left(\mu_0(x), \left(\frac{h_s}{h_T}\right)^2 \left(\frac{1}{h_0} + \frac{1}{h_s}\right) + \left(\frac{h_\eta}{h_T}\right)^2 \left(\frac{1}{h_0} + \frac{1}{h_\eta}\right) + 2\frac{h_s}{h_T} \frac{h_\eta}{h_T} \frac{1}{h_0}\right) \\ &\sim N\left(\mu_0(x), \frac{h_s^2 + h_\eta^2 + h_0^2 - h_0^2 + 2h_s h_\eta h_s h_0 + h_\eta h_0}{h_0 h_T^2}\right) \\ &\sim N\left(\mu_0(x), \frac{h_T - h_0}{h_0 h_T}\right) \\ &\sim N\left(\mu_0(x), \sigma_0^2 \rho_T^2\right). \end{split}$$

Further manipulation finally gives the hiring threshold:

$$\frac{\mu(x,\eta,s) - \mu_0(x)}{\sigma_0 \rho_T} > \underbrace{\frac{k_T - \frac{h_s}{h_T} \nu_s(x) - \frac{h_0 + h_\eta}{h_T} \nu_\eta(x) - \pi_j(x) (d_\eta + d_s) - \mu_0(x)}{\sigma_0 \rho_T}}_{z_T^*(x)}.$$
(C.3)

With two screening tools, evaluators place less weight on their group priors, which favors the hiring threshold of the minority group if $\mu_0(m) > \mu_0(f)$. Moreover, the overall bias now captures bias from both tools.

A.4 Hiring Rate With Blind Written Exam

Within the model, blinding makes it impossible to assign individual candidates to a group, since hiring evaluators cannot observe whether a certain signal is generated by a male or female candidate. A blind written exam provides the following signal:

$$s(b)^* = y + v_s(x) + \varepsilon_s, \ \varepsilon \sim N(0, 1/h_s)$$

with the same screening precision h_s and the same disparate impact $v_s(x)$ as the written test. When the screening technology includes blind written tests, the evaluator's objective becomes:

$$u(y, \pi(x)) = y + \underbrace{(1 - c_{s(b)})}_{d_{s(b)} = 0} \pi(x) \equiv y,$$

as discretion is entirely removed from the screening tool. Additionally, blinding the written test affects how the evaluator updates perceived candidate productivity, using the written signal, *s*, and the perceived *population* productivity, $\mu_0 = \frac{\mu_0(x) + \mu_0(y)}{2}$, since group membership is not identifiable:³⁶

$$\mu(x,s(b)^*) = s \frac{h_s}{h_0 + h_s} + \frac{\mu_0(x) + \mu_0(y)}{2} \frac{h_0}{h_0 + h_s} + \nu_s(x).$$

³⁶For simplicity and without loss of generality, I assume that each group comprises half of the candidate pool. Another reason for using identical gender distributions is to keep application behavior from the pool of qualified workers outside the model.

Manipulating the above gives the hiring threshold for group *x*:

$$\frac{s - \mu_0(x)}{\frac{1}{h_0} + \frac{1}{h_s}} > \frac{(h_0 + h_s)(k_{s(b)} - \nu_s(x) - \mu_0(x))}{h_s \sqrt{(\frac{1}{h_0} + \frac{1}{h_s})}} - \frac{h_0(\mu_0(y) - \mu_0(x))}{2h_s \sqrt{(\frac{1}{h_0} + \frac{1}{h_s})}}$$

and finally,

$$\frac{s - \mu_0(x)}{\frac{1}{h_0} + \frac{1}{h_s}} > \underbrace{\frac{k_{s(b)} - \nu_s(x) - \mu_0(x)}{\sigma_0 \rho_s} - \frac{h_0 \rho_s}{2h_s \sigma_0} (\mu_0(y) - \mu_0(x))}_{z_b^*(x)}.$$
(C.4)

The hiring threshold for group *x* determined by $s(b)^*$ is similar to the expression obtained for written test screening, s^* , with an important distinction. While the signal given by the blind written test is just as informative as in the non-blind case, now evaluators update a group-neutral prior.

A.5 Hiring Rate With Blind Written and Non-Written Exams

Lastly, consider blinding a written exam when the screening process also includes a non-written test. This is similar to the previous case of combining screening signals from both exams, except for the blind written exam having no disparate treatment. However, evaluators still rely on group means and express bias in the overall posterior because of the non-written signal. Given the two signals, η^* and $s(b)^*$, the posterior is:

$$y \mid_{\eta^*, s(b)^*} \sim N(\mu(x, \eta^*, s(b)^*), 1/h_T)$$

and the updated mean

$$\mu(x,\eta^*,s(b)^*) = \frac{h_s s + h_\eta \eta + h_0 \mu_0(x) + \nu_s(x) h_s + \nu_\eta(x)(h_0 + h_\eta)}{h_T}.$$

Since η and s can be rewritten as $\eta = y + \varepsilon_{\eta}$ and $s = y + \varepsilon_{s}$, and $y, \varepsilon_{s}, \varepsilon_{\eta}$ are independent, it follows that:

$$\mu(x,\eta,s(b)) \equiv \frac{h_s s + h_\eta \eta + h_0 \mu_0(x)}{h_T} \sim N\left(\mu_0(x), \sigma_0^2 \rho_T^2\right)$$

The hiring decision can then be rewritten as:

$$\frac{h_{s}s + h_{\eta}\eta + h_{0}\mu_{0}(x)}{h_{T}} > k_{\eta s(b)} - \nu_{s}(x)\frac{h_{s}}{h_{T}} - \nu_{\eta}(x)\frac{(h_{0} + h_{\eta})}{h_{T}} - d_{\eta}\pi(x)$$

$$\frac{\mu(x, \eta, b) - \mu_{0}(x)}{\sigma_{0}\rho_{T}} > \underbrace{\frac{k_{\eta s(b)} - \nu_{s}(x)\frac{h_{s}}{h_{T}} - \nu_{\eta}(x)\frac{h_{\eta} + h_{0}}{h_{T}} - d_{\eta}\pi(x) - \mu_{0}(x)}{z_{\eta s(b)}^{*}(x)}.$$
(C.5)

B Predictions for Gender Hiring Gaps

B.1 Change in Hiring Rate for $w \longrightarrow w(b)$

Without loss of generality, assume that female candidates are less productive on average than men: $\mu_0(f) < \mu_0(m)$, or in other words, that women are the minority group. By blinding the written exam, how do screening thresholds $z_s^*(x)$ and $z_b^*(x)$ compare and thus how are hiring rates affected? By inspecting expressions (C.1) and (C.4) and considering that written tests — whether blind or not — have the same screening precision and disparate impact, since they are otherwise identical save for hiding a candidate's identity, women face a lower hiring threshold in the blinded exam, $z_s^*(f) > z_b^*(m)$, if and only if

$$d_s \pi_j(f) < \frac{h_0 \rho_s^2}{2h_s} (\mu_0(m) - \mu_0(f)).$$

The expression above captures the following intuition. As long as the evaluator favors male candidates, either through statistical discrimination or evaluator bias, blinding the written exam increases hiring rates for women. The right-hand side is always positive since $\mu_0(f) < \mu_0(m)$, and it represents the improvement in women's hiring odds from removal of the ability to statistically discriminate. Therefore, if the left-hand side — which captures evaluator bias — is negative, i.e., if hiring managers favor men, or if it is sufficiently small due to either low discretion or low bias, then the hiring rate for women increases and the hiring rate for men decreases after blinding the written exam. Alternatively, if an evaluator is biased in favor of women, blinding the exam curbs the evaluator's ability to balance women's penalty from statistical discrimination with personal bias, potentially decreasing the female hiring rate.

B.2 Change in Hiring Rate for $nw \longrightarrow w(b)$

I now analyze the potential change induced by the policy that most dramatically alters the mix of screening tools. To build intuition, consider an employer that solely relies on interviews to screen candidates. From the expression in (C.2), the disparate impact of interviews, their precision, and how much they enable evaluator bias to be expressed all determine an applicant's hiring odds. Only in terms of evaluator bias, under the assumption that interviews offer more discretion than written exams, this pre-policy state contains the highest expression of evaluator bias. In contrast, as discussed before, screening solely based on written exams is likely to provide a setting with low disparate treatment.

Assume $h_s = h_\eta$ and $\mu_s = \mu_\eta$. It follows that the hiring threshold for men is higher with the blind-written signal than with the non-written signal, $z_{\eta}^*(m) < z_b^*(m)$, as long as evaluators favor men $\pi_j(m) > 0$ or, alternatively, if the following is satisfied

$$\frac{d_{\eta}\pi_{j}(m) + (k_{b} - k_{\eta})}{\sigma_{0}\rho} > \frac{h_{0}\rho}{2h_{s}\sigma_{0}}(\mu_{0}(f) - \mu_{0}(m)),$$

which allows for sufficiently small evaluator bias toward women. Because the above inequality implies a higher threshold for hiring male candidates under blind written compared to non-written screening, it increases selectivity for men, and, given a constant total hiring rate, *K*, the gender hiring gap decreases.

Next, conduct the same exercise but now allow for written and non-written exams to have different disparate impacts, $\Delta v_s \neq \Delta v_\eta$, where $\Delta v_s = v_s(m) - v_s(f)$. In this case, changing from non-written screening stages to blind-written exams increases female hiring rates if and only if:

$$\frac{h_0\rho}{h_s\sigma_0}(\mu_0(f) - \mu_0(m)) < \frac{d_\eta(\pi_j(m) - \pi_j(f)) + (\Delta\nu_\eta - \Delta\nu_s)}{\sigma_0\rho}$$
(C.6)

Note that the left-hand side of the expression above is negative, so that if evaluators are men-favoring and interviews have a larger disparate impact than written exams, the inequality is satisfied and female hiring rates increase. In other words, if the principal substitutes a hiring tool for one that has a smaller disparate impact and eliminates discretion, the change will raise hiring rates of the minority group. More generally, if either evaluator bias favors men, or if the relative bias of non-written tests is lower than that of written tests, it can still increase female hiring as long as it satisfies the inequality above. Another way to interpret the inequality (C.6) is to rewrite it as

$$\frac{h_0\rho}{h_s\sigma_0}\mu_0(f) + \frac{d_\eta\pi_j(f)}{\sigma_0\rho} + \frac{(\nu_\eta(f) - \nu_s(f))}{\sigma_0\rho} < \frac{h_0\rho}{h_s\sigma_0}\mu_0(m) + \frac{d_\eta\pi_j(m)}{\sigma_0\rho} + \frac{(\nu_\eta(m) - \nu_s(m))}{\sigma_0\rho}$$
(C.7)

The left-hand side represents the perceived productivity of female applicants, equal to true productivity plus bias, either from the evaluator or screening tool. The right-hand side represents the perceived productivity of male applicants. Thus, if female applicants are perceived as less productive under non-written screening relative to written screening, then the transition increases their hiring rate.

Finally, relax the assumption of identical screening precisions. If written tests are more precise, $h_s > h_\eta$, switching from non-written screening to written testing raises the hiring rate of the group with lower perceived productivity, that is, it raises the female hiring rate if (C.7) holds. However, if interviews have higher precision, $h_s < h_\eta$, the transition from interviews to written test decreases screening precision and leads to higher hiring rates of the favored group, men. The net effect then depends on the losses from decreased screening precision relative to the gains from lower bias if (C.7) is satisfied.

B.3 Change in Hiring Rate for $nw \rightarrow w(b) + nw$

This case maintains the use of non-written exams but, to comply with the impartiality policy, the employer adds a blind-written exam to the hiring process. By having an additional evaluation tool, total hiring precision increases, $h_0 + h_\eta + h_s > h_0 + h_\eta$, without introducing disparate treatment, since $d_b = 0$. Adding the blind-written tool reduces the weight that discretion in the non-written test plays in determining hiring rates (recall that $\frac{d_\eta \pi_i(x)}{\sigma_0 \rho_T} < \frac{d_\eta \pi_j(x)}{\sigma_0 \rho_\eta}$). However, introducing a different screening tool potentially incorporates that tool's disparate impact.

To start assume that screening tools do not favor any group, that is $\nu_{\eta}(f) = \nu_{\eta}(m)$, $\nu_{s}(f) = \nu_{s}(m)$, and that $\nu_{\eta} = \nu_{s}$. Then,

$$z_{\eta b}^{*}(x) = \frac{k_{\eta b} - \nu(x) - \mu_{0}(x) - d_{\eta}\pi_{j}(x)}{\sigma_{0}\rho_{T}} < \frac{k_{\eta} - \nu(x) - \mu_{0}(x) - d_{\eta}\pi_{j}(x)}{\sigma_{0}\rho_{\eta}} = z_{\eta}^{*}$$

That is, the hiring threshold is lower for group f if $\mu_0(f) + d_\eta \pi_j(f) < \mu_0(m) + d_\eta \pi_j(m)$ — women have lower perceived productivity. For the minority group both effects help as long as the same condition holds: $\mu_0(f) + d_\eta \pi_j(f) < \mu_0(m) + d_\eta \pi_j(m)$.

The increase in screening precision and decrease in relative importance of evaluator bias increase women's hiring rates if they are the group with the lower perceived productivity: $\mu_0(f) + d_\eta \pi_i(f) < \mu_0(m) + d_\eta \pi_i(m)$,

reflecting that the change in hiring probability with respect to screening precision is:

$$\frac{\partial \left[1 - \Phi(z_{\eta b}^*(x))\right]}{\partial \rho_T} = \phi(z_{\eta b}^*(x)) \left[\frac{z_{\eta b}^*(x)}{\rho_T} - \frac{\partial k_{\eta b} / \partial \rho_T}{\sigma_0 \rho_T}\right] > 0$$
(C.8)

Here $\phi(\cdot) > 0$ and $z_{\eta b}^*(m) < z_{\eta b}^*(f)$ if the above inequality of women being perceived as the group with lower productivity is satisfied.

Now, allow for screening tool bias to differ between written and non-written tests and to favor one group, $v_{\eta}(f) \neq v_{\eta}(m)$. Then, women benefit from the added precision if the following inequality holds:

$$\mu_0(f) + d_\eta \pi_j(f) + \nu_s(f) \frac{h_s}{h_T} + \nu_\eta(f) \frac{h_0 + h_\eta}{h_T} < \mu_0(m) + d_\eta \pi_j(m) + \nu_s(m) \frac{h_s}{h_T} + \nu_\eta(m) \frac{h_0 + h_\eta}{h_T}$$

If the written test that is added is bias increasing, $|\Delta v_s| > |\Delta v_\eta|$, it causes excess hiring of the group that is favored by the bias. Then, if the bias favors men, $v_s(m) - v_s(f) \equiv \Delta v_s > \Delta v_\eta \equiv v_\eta(m) - v_\eta(f)$, the net effect on the female hiring rate depends on the gains from increased screening precision relative to the losses from increased bias. On the other hand, if the bias favors women, and written tests are more biased than interviews, it leads unambiguously to higher hiring rates of women since all three forces have a positive effect.

B.4 Change in Hiring Rate for $w + nw \longrightarrow w(b)$

Removing the non-written signal from a screening mix of written and non-written decreases total screening precision, $h_0 + h_s < h_0 + h_s + h_{\eta}$, removes evaluator bias within the non-written test, $d_{\eta}\pi_j(x)$, and removes the non-written screening tool bias, $v_{\eta}(x)$. In addition, blinding the written test removes evaluator bias within the exam, $d_s\pi_j(x)$, as well as the use of group means (statistical discrimination) in determining the evaluator's posterior.

To begin with, assume $v_s = v_{\eta}$, which does not however eliminate the effect of removing the non-written screening tool bias, but just assumes that the type of tool bias reduced is the same in magnitude and sign (favors the same group), as the bias characterizing the written test.

Removing both screening tool and evaluator biases raises selectivity of the favored group and reduces selectivity of the non-favored group: $z_T^*(m) < z_h^*(m)$. Thus

$$\left[\frac{k_T - \nu(m) - \mu_0(m)}{\sigma_0 \rho_T} - \frac{k_b - \nu(m) - \mu_0(m)}{\sigma_0 \rho_s}\right] - \frac{\pi_j(m)(d_\eta + d_s)}{\sigma_0 \rho_T} + \frac{h_0 \rho_s}{2h_s \sigma_0}(\mu_0(f) - \mu_0(m)) < 0$$

where the inequality holds for *m* if this is the favored group. Thus, removing the non-written tool and evaluator bias, as well as evaluator bias within the written screening tool reduces selectivity of women and thus raises women hiring rates if they are the non-favored group.

However, the decrease in screening precision due to removal of the non-written signal has the opposite effect on hiring rates of the non-favored group:

$$\gamma_{f} \equiv \frac{\partial \left[1 - \Phi(z_{T}^{*}(f))\right]}{\partial \rho_{T}} = \phi(z_{T}^{*}(f)) \left[\frac{z_{T}^{*}(f)}{\rho_{T}} - \frac{\partial k / \partial \rho_{T}}{\sigma_{0} \rho_{T}}\right]$$

with ρ_T decreasing as $\rho_S < \rho_T$, $\phi(\cdot) > 0$, and $z_b^*(m) < z_b^*(f)$ if:

$$\frac{\mu_0(f) + \nu_s(f) - (\mu_0(m) + \nu_s(m))}{\sigma_0 \rho_s} + \frac{h_0 \rho_s}{h_s \sigma_0} (\mu_0(m) - \mu_0(f)) < 0$$

This later inequality holds if $\mu_0(f) + \nu_s(f) < \mu_0(m) + \nu_s(m)$ (men are the favored group, perceived to have higher productivity). Note that the inequality of men having the higher perceived productivity can hold even if the written test favors women, $\nu_s(f) > \nu_(m)$, if it is small enough: $\nu_s(f) - \nu_s(m) < \mu_0(m) - \mu_0(f)$. So with ρ decreasing, $\gamma_f < 0$ and $\gamma_m > 0$ if men are the favored group. Consequently, the net effect depends on the positive effect on female hiring rates from decreased bias relative to the negative effect from decreased screening precision.

Third, removing the non-written tool also eliminates its bias, v_{η} , which affects hiring rates depending on whether the bias favored men or women, as well as the relative size of this bias compared to the written tool bias. Consider the following cases.

Fist, suppose that the written tool favors women, $\Delta v_s < 0$, while the non-written favors men, $\Delta v_{\eta} > 0$, where $\Delta v_{\theta} = v_{\theta}(m) - v_{\theta}(f)$. Then, removing the non-written signal is bias-reducing and reduces excess hiring of the group favored by the non-written bias — men — increasing selectivity for the group and increasing the hiring rate for women. More formally, this follows from:

$$\begin{aligned} (z_T^*(m) - z_b^*(m)) - (z_T^*(f) - z_b^*(f)) &< 0\\ \frac{h_s \rho_s - h_T \rho_T}{\sigma_0 \rho_s \rho_T h_T} (\nu_s(f) - \nu_s(m)) + \frac{h_0 + h_\eta}{\sigma_0 \rho_T h_T} (\nu_\eta(f) - \nu_\eta(m)) &< 0\\ (\nu_\eta(f) - \nu_\eta(m)) &< \frac{h_T \rho_T - h_s \rho_s}{(h_0 + h_\eta) \rho_s} (\nu_s(f) - \nu_s(m)) \end{aligned}$$

where the fraction term is positive from $h_T = h_0 + h_\eta + h_s > h_s$. It follows that the right-hand side is also positive and the left-hand side is negative. This implies an increase in women's hiring rates.

Second, if instead the written signal favors men $\Delta v_s > 0$, while the non-written favors women, $\Delta v_{\eta} < 0$, then using the same inequality, it follows that removing the women-favoring bias from non-written increases women's selectivity, decreasing their hiring rate.

Third, if both the written and non-written tools favor men, $\Delta v_s > 0$, $\Delta v_{\eta} > 0$, then, regardless of which bias is larger, removing the non-written signal is bias-reducing and thus reduces excess hiring of the group favored by the bias, men, which in turn increases hiring rate for women. If, instead, both tools favor women, $\Delta v_s < 0$, $\Delta v_{\eta} < 0$, then, similarly, the transition is bias-reducing and decreases excess hiring of the favored group, which in this case are women. This increases selectivity for women, which decreases their hiring rate.